

Fast Kötter–Nielsen–Høholdt interpolation over skew polynomial rings and its application in coding theory

Hannes Bartz¹ · Thomas Jerkovits¹ · Johan Rosenkilde²

Received: 6 July 2022 / Revised: 21 September 2023 / Accepted: 4 October 2023 / Published online: 18 November 2023 © The Author(s) 2023

Abstract

Skew polynomials are a class of non-commutative polynomials that have several applications in computer science, coding theory and cryptography. In particular, skew polynomials can be used to construct and decode evaluation codes in several metrics, like e.g. the Hamming, rank, sum-rank and skew metric. We propose a fast divide-and-conquer variant of Kötter-Nielsen-Høholdt (KNH) interpolation algorithm: it inputs a list of linear functionals on skew polynomial vectors, and outputs a reduced Gröbner basis of their kernel intersection. We show, that the proposed KNH interpolation can be used to solve the interpolation step of interpolation-based decoding of interleaved Gabidulin codes in the rank-metric, linearized Reed-Solomon codes in the sum-rank metric and skew Reed-Solomon codes in the skew metric requiring at most $O(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{a^m} , where n is the length of the code, s the interleaving order, $\mathfrak{M}(n)$ the complexity for multiplying two skew polynomials of degree at most n, ω the matrix multiplication exponent and $\widetilde{O}(\cdot)$ the soft-O notation which neglects log factors. This matches the previous best speeds for these tasks, which were obtained by top-down minimal approximant bases techniques, and complements the theory of efficient interpolation over free skew polynomial modules by the bottom-up KNH approach. In contrast to the top-down approach the bottom-up KNH algorithm has no requirements on the interpolation points and thus does not require any pre-processing.

Communicated by J. Bierbrauer.

Part of this work was presented at the 25th International Symposium on Mathematical Theory of Networks and Systems (MTNS) [12].

Hannes Bartz hannes.bartz@dlr.de

Thomas Jerkovits thomas.jerkovits@dlr.de

Johan Rosenkilde jsrn@jsrn.dk

- ¹ Institute of Communications and Navigation, German Aerospace Center (DLR), 82234 Oberpfaffenhofen, Germany
- ² GitHub Inc., San Francisco, CA, USA

Keywords Fast Kötter interpolation · Skew polynomials · Linearized Reed-Solomon codes · Sum-rank metric · Skew metric

Mathematics Subject Classification 94B35 · 94B05

1 Introduction

Skew polynomials are a class of non-commutative polynomials, that were introduced by Ore in 1933 [53] and that have a variety of applications in computer science, coding theory and cryptography. The non-commutativity stems from the multiplication rule, which involves both, a field automorphism σ and a field derivation δ . Unlike ordinary polynomials, there exist several ways to evaluate skew polynomials. General results regarding the so-called *remainder evaluation* of skew polynomials were derived in [27, 29] whereas the *generalized operator evaluation* was considered in [31]. Depending on the choice of the automorphism σ and the derivation δ , skew polynomial rings (denoted by $\mathbb{F}_{q^m}[x; \sigma, \delta]$) include several interesting special cases, such as the ordinary polynomial ring as well as the linearized polynomial ring [52, 53]. This property along with the different ways to evaluate skew polynomials make them a very versatile tool with many different applications.

One important application of skew polynomials is the construction of evaluation codes, that have distance properties in several decoding metrics, including the *Hamming*, rank, sum-rank, skew and other related metrics such as the (sum-)subspace metric [14, 15, 39, 43].

Many evaluation codes allow for decoding via efficient interpolation-based decoding algorithms, like e.g. the Welch–Berlekamp [64] and Sudan [61] algorithms for decoding Reed–Solomon codes. In [26] Kötter presented a bivariate interpolation algorithm for Sudan-like decoding of Reed–Solomon codes [26] (over ordinary polynomial rings) that since then is often referred to as the *Kötter interpolation*. The Kötter interpolation as it is known today was first stated by Nielsen and Høholdt [50] as a generalization of Kötter's algorithm [26] which is able to handle multiplicities. To acknowledge the contribution by Nielsen and Høholdt we refer to the algorithm as Kötter–Nielsen–Høholdt (KNH) interpolation. A fast divide-and-conquer variant of the KNH interpolation for the Guruswami–Sudan algorithm for decoding Reed–Solomon codes was presented in [49]. Rosenkilde's algorithm [49] is a bottom-up KNH-like algorithm whose complexity is only slightly larger compared to the currently fastest approach [24, Sect. 2.5].

A multivariate generalization of the KNH interpolation for free modules over ordinary polynomial rings was proposed in [63]. This approach was generalized to free modules over linearized polynomial rings in [65]. A generalization of the multivariate KNH interpolation to free modules over skew polynomial rings was proposed in [33], which contains the variants over ordinary polynomial rings [63] and linearized polynomial rings [65] as a special case.

The evaluation and interpolation of multivariate skew polynomials was also considered in [42]; here with the main motivation to construct Reed-Muller-like codes (see also [4, 21, 41]).

1.1 Main contribution

In this paper, we propose a fast divide-and-conquer (D&C) variant of the KNH interpolation in skew polynomial rings [33], that uses ideas from [49]. The main idea of the proposed algorithm (Algorithm 3) is, that the interpolation problem is divided into smaller sub-problems,

that can be solved and merged efficiently. In particular, the update operations in each loop of the KNH interpolation are "recorded" and then applied to a degree-reduced basis in the merge step rather than to a non-reduced basis. This allows to control the degree of the polynomials during the interpolation procedure which in turn results in a lower computational complexity.

We state the interpolation problem and the algorithm in a general way using linear functionals over skew polynomials rings with arbitrary automorphisms and derivations. We show how the fast KNH interpolation can be applied to interpolation-based decoding of (interleaved) Gabidulin codes [35, 54], interleaved linearized Reed–Solomon codes [5, 6, 15, 39] and (interleaved) skew Reed–Solomon codes [5, 6, 14].

We consider skew polynomials over finite fields only. However, the results (except for the complexity statements) also hold for skew polynomials over arbitrary finite Galois extensions \mathbb{L}/\mathbb{K} instead of $\mathbb{F}_{q^m}/\mathbb{F}_q$ and automorphisms $\sigma \in \text{Gal}(\mathbb{L}/\mathbb{K})$ with $\mathbb{K} = \mathbb{L}^{\sigma}$ and derivations $\delta : \mathbb{K} \mapsto \mathbb{K}$ satisfying (2) for all $a, b \in \mathbb{K}$.

For the above mentioned applications using generalized operator and remainder evaluation maps over skew polynomial rings with arbitrary field automorphisms we discuss the asymptotic complexity for zero derivations ($\delta = 0$). The asymptotic complexity for solving the interpolation step with the proposed approach is $\tilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} , where *n* is the length of the code, *s* a decoding parameter (e.g. interpolation order, usually $s \ll n$), $\mathfrak{M}(n) \in \tilde{O}(n^{1.635})$ the complexity for multiplying two skew polynomials of degree at most *n*, ω the matrix multiplication exponent (currently $\omega < 2.37286$) and $\tilde{O}(\cdot)$ denotes the soft-*O* notation which neglects log factors.

The original skew KNH interpolation from [33] has an asymptotic complexity of $O(s^2n^2)$ operations in \mathbb{F}_{q^m} , which is larger compared to the proposed approach for most practical cases where we usually have $s \ll n$.

The interpolation step of the above mentioned coding applications can also be solved using the skew minimal approximant bases methods from [11] requiring at most $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} . This approach can be seen as a top–down approach: first construct a module that contains all solutions to the interpolation problem, and then find the minimal solution in that module. The KNH family of algorithms are bottom-up: Gradually build up a minimal basis solving the interpolation constraints one by one. Developing both approaches in tandem has been very fruitful for the analogous family of algorithms for ordinary polynomial rings.

Due to the top-down nature, the minimal approximant bases method [11, Algorithm 6] requires an additional step if the first entries of the interpolation points (related to the generalized operator evaluation maps) are not linearly independent (see [11, Theorem 22]), which is *not* required in the proposed KNH-like algorithm.

A comparison between the proposed fast KNH interpolation algorithm and existing interpolation methods for ordinary and skew polynomial rings, including the computational complexity for the zero-derivation case ($\delta = 0$), is given in Table 1. Note that for ordinary polynomial rings we have that $\mathfrak{M}(n) \in \widetilde{O}(n)$. Additionally, in the ordinary polynomial ring case we have the notion of multiplicities of roots; a concept which is not yet generalized to the skew polynomial setting and therefore omitted.

Although the proposed interpolation algorithm achieves the best known complexity over skew polynomial rings, Table 1 shows that the best computational complexity of $\widetilde{O}(s^{\omega-1}\mathfrak{M}(n))$ for ordinary polynomial rings [24] is not yet reached for the skew polynomial case. Closing this gap is a potential topic for future work.

Interpolation method		Туре	Polynomial ring	Complexity $(\delta = 0)$
Ordinary	KNH [63]	bottom-up	$\mathbb{F}_{q^m}[x; Id, 0]$	$O\left(s^2n^2\right)$
	D&C KNH [49]	bottom-up	$\mathbb{F}_{q^m}[x; Id, 0]$	$\widetilde{O}\left(s^{\omega}\mathfrak{M}(n)\right)$
	Min. approximant bases method [22]	top-down	$\mathbb{F}_{q^m}[x; Id, 0]$	$\widetilde{O}\left(s^{\omega}\mathfrak{M}(n)\right)$
	Min. interpolation bases [24]	top-down	$\mathbb{F}_{q^m}[x;Id,0]$	$\widetilde{O}\left(s^{\omega-1}\mathfrak{M}(n)\right)$
Skew	Linearized KNH [65]	bottom-up	$\mathbb{F}_{q^m}[x; \sigma_{\text{Frob}}, 0]$	$O\left(s^2n^2\right)$
	Skew KNH [33]	bottom-up	$\mathbb{F}_{q^m}[x;\sigma,\delta]$	$O\left(s^2n^2\right)$
	D&C skew KNH (this contribution)	bottom-up	$\mathbb{F}_{q^m}[x;\sigma,\delta]$	$\widetilde{O}\left(s^{\omega}\mathfrak{M}(n)\right)$
	Skew min. approximant bases method [11]	top-down	$\mathbb{F}_{q^m}[x;\sigma,\delta]$	$\widetilde{O}\left(s^{\omega}\mathfrak{M}(n)\right)$

Table 1 Overview of the computational complexity of the proposed fast KNH interpolation approach compared to existing methods for the case of zero derivations ($\delta = 0$)

Here *n* is the number of interpolation points, *s* an interpolation parameter (usually $s \ll n$), $\mathfrak{M}(n) \in \widetilde{O}(n^{1.635})$ the complexity for multiplying two skew polynomials of degree at most *n* and ω the matrix multiplication exponent (currently $\omega < 2.37286$). For ordinary polynomial rings we have that $\mathfrak{M}(n) \in \widetilde{O}(n)$

1.2 Outline of the paper

Section 2 gives definitions and notations related to skew polynomials as well as a definition of the skew KNH interpolation algorithm from [33]. Section 3 presents a fast general D&C framework for the skew KNH interpolation. Section 4 considers the application of the fast skew KNH interpolation for decoding (interleaved) Gabidulin, linearized Reed–Solomon and Skew Reed–Solomon codes. The complexity analysis shows that we obtain the currently fastest known decoders for the considered codes. Section 5 concludes the paper.

2 Preliminaries

2.1 Sets, vectors and matrices over finite fields

Let \mathbb{F}_q be a finite field and denote by \mathbb{F}_{q^m} the extension field of degree *m*. Sets are denoted by $\mathcal{A} = \{a_0, a_1, \dots, a_{n-1}\}$. The cardinality of a set \mathcal{A} is denoted by $|\mathcal{A}|$.

Vectors and matrices over \mathbb{F}_{q^m} are denoted by bold lower-case and upper-case letters such as **a** and **A**, respectively, and the elements are indexed beginning from zero. The $r \times c$ zero matrix is denoted by $\mathbf{0}_{r \times c}$ and the $r \times r$ identity matrix is denoted by \mathbf{I}_r . The *i*-th row of a matrix **A** is denoted by \mathbf{a}_i . Let $\mathbb{F}_{q^m}^N$ denote the set of all row vectors of length N over \mathbb{F}_{q^m} and let $\mathbb{F}_{q^m}^{M \times N}$ denote the set of all $M \times N$ matrices over \mathbb{F}_{q^m} . The rank of a matrix $\mathbf{A} \in \mathbb{F}_{q^m}^{M \times N}$ is denoted by $\mathbf{rk}_q(\mathbf{A})$. Under a fixed basis of \mathbb{F}_{q^m} over \mathbb{F}_q , there is a bijection between $\mathbb{F}_{q^m}^n$ and $\mathbb{F}_q^{m \times n}$. This allows us to define the \mathbb{F}_q -rank of a vector $\mathbf{a} \in \mathbb{F}_{q^m}^N$ as $\mathbf{rk}_q(\mathbf{a}) := \mathbf{rk}(\mathbf{A})$ where \mathbf{A} is the corresponding matrix of \mathbf{a} over \mathbb{F}_q . Note that the \mathbb{F}_q -rank $\mathbf{rk}_q(\mathbf{a})$ is independent of the choice of the basis of \mathbb{F}_{q^m} over \mathbb{F}_q and can be equivalently seen as the maximum number of

 \mathbb{F}_q -linearly independent entries of **a**. The *Hamming* weight of a vector $\mathbf{a} \in \mathbb{F}_{q^m}^N$ is defined as

$$wt_H(\mathbf{a}) = |\{i \in \{1, \dots, N\} : a_i \neq 0\}|.$$
(1)

Let $\sigma : \mathbb{F}_{q^m} \mapsto \mathbb{F}_{q^m}$ be a field automorphism of \mathbb{F}_{q^m} and let $\delta : \mathbb{F}_{q^m} \mapsto \mathbb{F}_{q^m}$ be a σ -derivation such that

$$\delta(a+b) = \delta(a) + \delta(b)$$
 and $\delta(ab) = \delta(a)b + \sigma(a)\delta(b)$. (2)

Over a finite field, all σ -derivations are of the form (see e.g. [33, Proposition 1])

$$\delta(a) = b(\sigma(a) - a) \quad \text{for } b \in \mathbb{F}_{q^m}.$$
(3)

From (3) we see that we get the zero derivation ($\delta = 0$) if the automorphism is the identity ($\sigma = Id$).

For any two elements $a \in \mathbb{F}_{q^m}$ and $c \in \mathbb{F}_{q^m}^* = \mathbb{F}_{q^m} \setminus \{0\}$ define

$$a^{c} := \sigma(c)ac^{-1} + \delta(c)c^{-1}$$
(4)

where term $\delta(c)c^{-1}$ is called the *logarithmic derivative* of c. Two elements $a, b \in \mathbb{F}_{q^m}$ are called (σ, δ) -conjugates, if there exists an element $c \in \mathbb{F}_{q^m}^*$ such that $b = a^c$. Otherwise, a and b are called (σ, δ) -distinct. The notion of (σ, δ) -conjugacy defines an equivalence relation on \mathbb{F}_{q^m} and thus a partition of \mathbb{F}_{q^m} into conjugacy classes (see [29]).

Definition 1 (Conjugacy Class [29]) The set

$$\mathcal{C}(a) := \left\{ a^c : c \in \mathbb{F}_{q^m}^* \right\}$$
(5)

is called *conjugacy class* of a.

2.2 Skew polynomials

Skew polynomials are non-commutative polynomials that were introduced by Ore [53]. The set of all polynomials of the form

$$f(x) = \sum_{i} f_{i} x^{i} \quad \text{with} \quad f_{i} \in \mathbb{F}_{q^{m}}$$
(6)

together with the ordinary polynomial addition and the multiplication rule

$$xa = \sigma(a)x + \delta(a) \tag{7}$$

forms the non-commutative ring of *skew polynomials* that is denoted by $\mathbb{F}_{q^m}[x; \sigma, \delta]$. The *degree* of a skew polynomial $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ is defined as $deg(f) := \max_i \{i : f_i \neq 0\}$ for $f \neq 0$ and $-\infty$ else. Further, by $\mathbb{F}_{q^m}[x; \sigma, \delta]_{< n}$ we denote the set of skew polynomials from $\mathbb{F}_{q^m}[x; \sigma, \delta]$ of degree less than n.

For δ being an inner derivation (3), which is always the case for finite fields, there is a ring isomorphism between $\mathbb{F}_{q^m}[x; \sigma, \delta]$ and $\mathbb{F}_{q^m}[x; \sigma]$ that is given by the mapping (see [39, Proposition 40] and [32, Proposition 2.1.8])

$$\mathbb{F}_{q^m}[x;\sigma,\delta] \to \mathbb{F}_{q^m}[x;\sigma]$$
$$\sum_i f_i x^i \mapsto \sum_i f_i (x+b)^i.$$

Deringer

The *monic* least common left multiple (LCLM) of some polynomials $p_0, p_1, \ldots, p_{n-1} \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ is denoted by

$$\operatorname{lclm}(p_i)_{0 \le i \le n-1} := \operatorname{lclm}(p_0, p_1, \dots, p_{n-1}).$$
(8)

The skew polynomial ring $\mathbb{F}_{q^m}[x; \sigma, \delta]$ is a left and right Euclidean domain, i.e., for any $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ and any nonzero $g \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ there exist unique polynomials $q_L, r_L, q_R, r_R \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ such that

$$f(x) = q_R(x)g(x) + r_R(x) = g(x)q_L(x) + r_L(x)$$
(9)

where $\deg(r_R)$, $\deg(r_L) < \deg(g)$ (see [53]). Efficient Euclidean-like algorithms for performing left/right skew polynomial division exist [16, 17, 55]. For two skew polynomials $f, g \in \mathbb{F}_{q^m}[x; \sigma, \delta]$, denote by $f \mod_r g$ the remainder of the right division of f by g.

Example 1 Applying the multiplication rule in (7) to x^2a we get

$$\begin{aligned} x^2 a &= x(\sigma(a)x + \delta(a)) = x\sigma(a)x + x\delta(a) \\ &= (\sigma^2(a)x + \delta(\sigma(a)))x + \sigma(\delta(a))x + \delta^2(a) \\ &= \sigma^2(a)x^2 + (\delta(\sigma(a)) + \sigma(\delta(a)))x + \delta^2(a). \end{aligned}$$

There are several interesting cases where skew polynomial rings coincide with other polynomial rings:

- For σ being the identity we have that δ is the zero derivation (see (3)) which implies that $\mathbb{F}_{q^m}[x; \sigma, \delta]$ is equivalent to the ordinary polynomial ring $\mathbb{F}_{q^m}[x]$.
- For δ being the zero derivation we get the *twisted* polynomial ring $\mathbb{F}_{q^m}[x; \sigma]$.
- For σ_{Frob} being the Frobenius automorphism of \mathbb{F}_{q^m} (i.e. $\sigma_{\text{Frob}}(\cdot) = \cdot^q$) and δ being the zero derivation we have that $\mathbb{F}_{q^m}[x; \sigma_{\text{Frob}}]$ is isomorphic to the linearized polynomial ring $\mathbb{L}_{q^m}[x]$ [52, 53].

There exist two variants of skew polynomial evaluation: the *(generalized) operator* evaluation and the *remainder* evaluation.

2.2.1 Generalized operator evaluation

The generalized operator evaluation defined in [31] allows to \mathbb{F}_q -linearize the skew polynomial evaluation and therefore establishes the link between the skew polynomial ring and the linearized polynomial ring [52, 53].

Given an \mathbb{F}_{q^m} automorphism σ , a σ -derivation δ and an element $a \in \mathbb{F}_{q^m}$, the (σ, δ) operator $\mathcal{D}_a^{\sigma,\delta}(b) : \mathbb{F}_{q^m} \mapsto \mathbb{F}_{q^m}$ is defined as

$$\mathcal{D}_{a}^{\sigma,\delta}(b) := \sigma(b)a + \delta(b), \quad \forall b \in \mathbb{F}_{q^{m}}.$$
(10)

We use the notation $\mathcal{D}_a(b)$ whenever σ and δ are clear from the context. For an integer $i \ge 0$, we define $\mathcal{D}_a^{i+1}(b) = \mathcal{D}_a(\mathcal{D}_a^i(b))$ and $\mathcal{D}_a^0(b) = b$.

Definition 2 (*Generalized Operator Evaluation* [39]) For a skew polynomial $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ the generalized operator evaluation $f(b)_a$ of f at an element $b \in \mathbb{F}_{q^m}$ w.r.t. the evaluation parameter $a \in \mathbb{F}_{q^m}$ is defined as

$$f(b)_a := \sum_i f_i \mathcal{D}_a^i(b).$$
(11)

The definition of the generalized operator evaluation includes the operator evaluation $f(b)_1$ as a special case (a = 1).

The generalized operator evaluation is an \mathbb{F}_q -linear map, i.e. for any $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$, $\lambda_1, \lambda_2 \in \mathbb{F}_q$ and $a, b_1, b_2 \in \mathbb{F}_{q^m}$ we have that (see [39, Lemma 23] and [30])

$$f(\lambda_1 b_1 + \lambda_2 b_2)_a = \lambda_1 f(b_1)_a + \lambda_2 f(b_2)_a.$$
 (12)

For a vector $\mathbf{b} = (b_0, b_1, \dots, b_{n-1}) \in \mathbb{F}_{q^m}^n$ we define the generalized multipoint operator evaluation of a skew polynomial $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ w.r.t. an $a \in \mathbb{F}_{q^m}$ as

$$f(\mathbf{b})_a := (f(b_0)_a, f(b_1)_a, \dots, f(b_{n-1})_a).$$
(13)

For a tuple $\mathcal{B} = (b_0, b_1, \dots, b_{n-1}) \in \mathbb{F}_{q^m}^n$ and a vector $\mathbf{a} = (a_0, a_1, \dots, a_{n-1}) \in \mathbb{F}_{q^m}^n$ the minimal skew polynomial that vanishes on all elements in \mathcal{B} w.r.t. the evaluation parameters in \mathbf{a} is defined as (see e.g. [15])

$$M_{\mathcal{B}}^{\text{op}}(x)_{\mathbf{a}} = \operatorname{lclm}\left(x - \frac{\sigma(b_i)a_i + \delta(b_i)}{b_i}\right)_{\substack{0 \le i \le n-1\\b_i \ne 0}}.$$
(14)

The degree of $M_{\mathcal{B}}^{\text{op}}(x)_{\mathbf{a}}$ satisfies

$$\deg(M_{\mathcal{B}}^{\mathrm{op}}(x)_{\mathbf{a}}) \le n \tag{15}$$

where equality holds if the sequence of elements b_i that have the same evaluation parameter a_i are \mathbb{F}_q -linearly independent and the distinct evaluation parameters a_i are from different conjugacy classes (see [15]).

Example 2 Consider the tuple $\mathcal{B} = (b_0, b_1, b_2, b_3) \in \mathbb{F}_{q^m}^4$ and a vector $\mathbf{a} = (a_0, a_1, a_2, a_3) \in \mathbb{F}_{q^m}^4$, where $a_0 = a_1$ and $a_2 = a_3$ are representatives from different conjugacy classes. Then we have deg $(M_{\mathcal{B}}^{op}(x)_{\mathbf{a}}) = n$ if and only if b_0 and b_1 are \mathbb{F}_q -linearly independent and b_2 and b_3 are \mathbb{F}_q -linearly independent.

Similar to ordinary polynomials, we get the following result for the generalized operator evaluation of a polynomial modulo a particular minimal polynomial.

Lemma 1 For any $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$, $\mathcal{B} = (b_0, b_1, \dots, b_{n-1}) \in \mathbb{F}_{q^m}^n$ and $\mathbf{a} = (a_0, a_1, \dots, a_{n-1}) \in \mathbb{F}_{q^m}^n$ we have that

$$f(b_i)_{a_i} = \left(f(x) \bmod_{\mathbf{r}} M^{op}_{\mathcal{B}}(x)_{\mathbf{a}}\right)(b_i)_{a_i}, \quad \forall i = 0, \dots, n-1.$$

$$(16)$$

Proof Since $\mathbb{F}_{q^m}[x; \sigma, \delta]$ is a left/right Euclidean domain, there exist two unique polynomials $q, r \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ with $\deg(r(x)) < \deg(M_{\mathcal{B}}^{op}(x)_{\mathbf{a}})$ such that

$$f(x) = q(x)M_{\mathcal{B}}^{\text{op}}(x)_{\mathbf{a}} + r(x) \quad \Longleftrightarrow \quad r(x) = f(x) \mod_{\mathbf{r}} M_{\mathcal{B}}^{\text{op}}(x)_{\mathbf{a}}.$$
(17)

Since $M_{\mathcal{B}}^{\text{op}}(x)_{\mathbf{a}}$ vanishes on all b_i w.r.t. a_i for all i = 0, ..., n - 1 we have that

$$f(b_i)_{a_i} = r(b_i)_{a_i}, \quad \forall i = 0, \dots, n-1,$$
 (18)

and the result follows.

Springer

2.2.2 Remainder evaluation

Another variant of skew polynomial evaluation is the *remainder evaluation* defined in [27, 29], which generalizes the concept of polynomial evaluation by means of (right) division.

Definition 3 (*Remainder Evaluation* [27, 29]) For a skew polynomial $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ the remainder evaluation f[b] of f at an element $b \in \mathbb{F}_{q^m}$ is defined as the unique remainder of the right division of f(x) by (x - b) such that

$$f(x) = g(x)(x-b) + f[b] \iff f[b] = f(x) \operatorname{mod}_{r} (x-b).$$
(19)

For a vector $\mathbf{b} = (b_0, b_1, \dots, b_{n-1}) \in \mathbb{F}_{q^m}^n$ we define the multipoint remainder evaluation of a skew polynomial $f \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ as

$$f[\mathbf{b}] := \left(f[b_0], f[b_1], \dots, f[b_{n-1}] \right).$$
(20)

In the following we recall important properties of the concept of *P*-independence (or *polynomial* independence) from [27, 28, 39]. Given a set $\mathcal{F} \subseteq \mathbb{F}_{q^m}[x; \sigma, \delta]$ we define its *zero* set $\mathcal{Z}(\mathcal{F}) \subseteq \mathbb{F}_{q^m}$ as

$$\mathcal{Z}(\mathcal{F}) := \{ a \in \mathbb{F}_{q^m} : f[a] = 0, \forall f \in \mathcal{F} \}.$$
(21)

For a set $\mathcal{B} = \{b_0, b_1, \dots, b_{n-1}\} \subseteq \mathbb{F}_{q^m}$ we define its *associated ideal* as (see [39])

$$I(\mathcal{B}) = \{ f \in \mathbb{F}_{q^m}[x; \sigma, \delta] : f[b] = 0, \forall b \in \mathcal{B} \}.$$
(22)

For a set $\mathcal{B} = \{b_0, b_1, \dots, b_{n-1}\} \subseteq \mathbb{F}_{q^m}$ the unique minimal skew polynomial that vanishes on all elements in \mathcal{B} w.r.t. the remainder evaluation is defined as (see e.g. [14])

$$M_{\mathcal{B}}^{\text{rem}}(x) = \text{lclm}\,(x - b_i)_{0 \le i \le n-1}\,.$$
(23)

Since $M_{\mathcal{B}}^{\text{rem}}(x)$ has the minimal degree among all polynomials in $\mathbb{F}_{q^m}[x; \sigma, \delta]$ that vanish on \mathcal{B} it generates the *left* $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -ideal $I(\mathcal{B})$. The degree of $M_{\mathcal{B}}^{\text{rem}}(x)$ satisfies

$$\deg(M_{\mathcal{B}}^{\operatorname{rem}}(x)) \le n \tag{24}$$

where the elements $b_0, b_1, \ldots, b_{n-1}$ are called *P*-independent (or *polynomially independent*) if and only if deg($M_B^{\text{rem}}(x)$) = *n*.

The *closure* of a set $\mathcal{B} \subseteq \mathbb{F}_{q^m}$ is defined as the zero set of its minimal polynomial, i.e. as

$$\overline{\mathcal{B}} := \mathcal{Z}(I(\mathcal{B})) = \mathcal{Z}(M_{\mathcal{B}}^{\text{rem}}(x))$$
(25)

where \mathcal{B} is called *P*-closed if and only if $\overline{\mathcal{B}} = \mathcal{B}$. For a *P*-closed set \mathcal{B} it can be shown that any root of $M_{\mathcal{B}}^{\text{rem}}(x)$ is an (σ, δ) -conjugate of an element in \mathcal{B} (see [28]).

Similar to the result w.r.t. to the generalized operator evaluation in Lemma 1, we obtain the following result w.r.t. the remainder evaluation.

Lemma 2 For any
$$p \in \mathbb{F}_{q^m}[x; \sigma, \delta]$$
 and $\mathcal{B} = \{b_0, b_1, \dots, b_{n-1}\} \subseteq \mathbb{F}_{q^m}$ we have that

$$p[b_i] = (p(x) \mod_{\mathbf{r}} M_{\mathcal{B}}^{rem}(x))[b_i], \quad \forall i = 0, \dots, n-1.$$
(26)

Proof Since $\mathbb{F}_{q^m}[x; \sigma, \delta]$ is a left/right Euclidean domain, there exist two unique polynomials $q, r \in \mathbb{F}_{q^m}[x; \sigma, \delta]$ with $\deg(r(x)) < \deg(M_B^{rem}(x))$ such that

$$p(x) = q(x)M_{\mathcal{B}}^{\text{rem}}(x) + r(x) \quad \Longleftrightarrow \quad r(x) = p(x) \mod_{r} M_{\mathcal{B}}^{\text{rem}}(x).$$
(27)

Since $M_{\mathcal{B}}^{\text{rem}}(x)$ vanishes on all $b_i \in \mathcal{B}$ we have that $p[b_i] = r[b_i]$ for all i = 0, ..., n-1 and the result follows.

The following result from [31] (see also [39]) shows the relation between the generalized operator and the remainder evaluation.

Lemma 3 (Connection between Evaluation Types [31, 39]) For any $a \in \mathbb{F}_{q^m}$, $b \in \mathbb{F}_{q^m}^*$ and $f \in \mathbb{F}_{q^m}[x; \sigma]$ we have that

$$f\left[\mathcal{D}_a(b)b^{-1}\right]b = f(b)_a.$$
(28)

2.2.3 Skew polynomial vectors and matrices

To be consistent with the conventional notation in coding theory we denote both, vectors and matrices, by bold letters. The dimensions are clear from the context.

For two vectors $\mathbf{a}, \mathbf{b} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^n$ we denote the element-wise right modulo operation by

$$\mathbf{a} \mod_{\mathbf{r}} \mathbf{b} := (a_0 \mod_{\mathbf{r}} b_0, a_1 \mod_{\mathbf{r}} b_1, \dots, a_{n-1} \mod_{\mathbf{r}} b_{n-1}).$$
(29)

For two vectors $\mathbf{a}, \mathbf{b} \in \mathbb{F}_{a^m}[x; \sigma, \delta]^n$ we define the element-wise LCLM as

$$\operatorname{lclm}(\mathbf{a}, \mathbf{b}) := \left(\operatorname{lclm}(a_0, b_0), \operatorname{lclm}(a_1, b_1), \dots, \operatorname{lclm}(a_{n-1}, b_{n-1})\right).$$
(30)

For a vector $\mathbf{a} = (a_0, a_1, \dots, a_{n-1}) \in \mathbb{F}_{q^m}[x; \sigma, \delta]^n$ and a vector $\mathbf{w} = (w_0, w_1, \dots, w_{n-1}) \in \mathbb{Z}_+^n$ we define its w-weighted degree as

$$\deg_{\mathbf{w}}(\mathbf{a}) := \max_{0 \le j \le n-1} \{\deg(a_j) + w_j\}.$$
(31)

Further, we define the w-weighted monomial ordering $\prec_{\mathbf{w}}$ on $\mathbb{F}_{q^m}[x; \sigma, \delta]^n$ such that we have

$$b_{\ell} x^{\ell} \mathbf{e}_{j} \prec_{\mathbf{w}} b_{\ell'} x^{\ell'} \mathbf{e}_{j'}$$
(32)

for $b_{\ell}, b_{\ell'} \in \mathbb{F}_{q^m} \setminus \{0\}$ if $\ell + w_j < \ell' + w_{j'}$ or if $\ell + w_j = \ell' + w_{j'}$ and j < j', where \mathbf{e}_j denotes the *j*-th unit vector over $\mathbb{F}_{q^m}[x; \sigma, \delta]$. The definition of $\prec_{\mathbf{w}}$ coincides with the **w**-weighted term-over-position (TOP) ordering as defined in [1].

For a vector $\mathbf{a} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^n \setminus \{\mathbf{0}\}$ and weighting vector $\mathbf{w} = (w_0, \dots, w_{n-1}) \in \mathbb{Z}_+^n$, we define the **w**-pivot index $\operatorname{Ind}_{\mathbf{w}}(\mathbf{a})$ of **a** to be the largest index j with $0 \le j \le n-1$ such that $\deg(a_j) + w_j = \deg_{\mathbf{w}}(\mathbf{a})$.

For a nonzero vector $\in \mathbb{F}_{q^m}[x; \sigma, \delta]^n$ we identify the *leading term* LT(**a**) of **a** as the maximum term $a_{i,j}x^j$ under $\prec_{\mathbf{w}}$. Note that in this case *j* coincides with the **w**-pivot index of **a**.

A matrix $\mathbf{A} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{a \times b}$ with $a \le b$ is in (row) w-ordered weak Popov form if the w-pivot indices of its rows are strictly increasing in the row index [46].

A free $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -module is a module that has a basis that consists of $\mathbb{F}_{q^m}[x; \sigma, \delta]$ linearly independent elements. The rank of this module equals the cardinality of that basis. In the following we consider particular bases for (left) $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -modules.

Definition 4 (w-ordered weak-Popov Basis [11]) Consider a left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodule \mathcal{M} of $\mathbb{F}_{q^m}[x; \sigma, \delta]^b$. For $\mathbf{w} \in \mathbb{Z}^a$, a left w-ordered weak-Popov basis for \mathcal{M} is a full-rank matrix $\mathbf{A} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{a \times b}$ s.t.

- 1. A is in w-ordered weak Popov form.
- 2. The rows of \mathbf{A} are a basis of \mathcal{M} .

We will now establish a connection between **w**-ordered weak-Popov Bases and Gröbner bases w.r.t. $\prec_{\mathbf{w}}$ for left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodules. This connection is well-known for ordinary commutative polynomial rings (see e.g. [2, 19, 25, 47, 48]). For skew polynomial rings this relation was derived in [45, Chapter 6] and also used in [11].

For a short introduction to Gröbner bases the reader is referred to [60]. An extensive study of Gröbner bases can be found in [18].

Definition 5 (Gröbner *Basis* [18]) Let \mathcal{M} be a left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodule. A subset $\mathcal{B} = \{\mathbf{b}_0, \mathbf{b}_1, \dots, \mathbf{b}_{\nu-1}\} \subset \mathcal{M}$ is called a Gröbner basis for \mathcal{M} under $\prec_{\mathbf{w}}$ if the leading terms of \mathcal{B} span a left module that contains all leading terms in \mathcal{M} , i.e. if $\langle \text{LT}(b_0), \dots, \text{LT}(b_{\nu-1}) \rangle = \langle \text{LT}(\mathcal{M}) \rangle$.

A Gröbner basis for a $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodule \mathcal{M} is not necessarily a *minimal* generating set for \mathcal{M} since any subset of \mathcal{M} that contains a Gröbner basis is also a Gröbner basis (see [18, 60]). The following definition imposes a minimality requirement on the cardinality of Gröbner bases for an $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodule under $\prec_{\mathbf{w}}$.

Definition 6 (*Minimal Gröbner Basis* [18]) Given a monomial ordering $\prec_{\mathbf{w}}$, a Gröbner basis \mathcal{B} for a left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodule \mathcal{M} is called minimal if for all $\mathbf{p} \in \mathcal{B}$ the leading term LT(\mathbf{p}) is not contained in the module $\langle LT(\mathcal{B} \setminus \{\mathbf{p}\}) \rangle$, i.e. if LT(\mathbf{p}) $\notin \langle LT(\mathcal{B} \setminus \{\mathbf{p}\}) \rangle$.

A minimal Gröbner basis \mathcal{B} w.r.t. to $\prec_{\mathbf{w}}$ is called *reduced* Gröbner basis if all leading terms are normalized and no monomial of $\mathbf{p} \in \mathcal{B}$ is in $\langle LT(\mathcal{B} \setminus \{\mathbf{p}\}) \rangle$.

Although [45, Theorem 6.29] establishes the connection between the stronger w-ordered Popov form and the corresponding *reduced* Gröbner basis w.r.t. \prec_w , the arguments also hold for the relation between the w-ordered weak-Popov form and the minimal Gröbner basis w.r.t. \prec_w .

Note that given a module monomial order \prec and a basis $\mathcal{B} \subset \mathbb{F}_{q^m}[x; \sigma, \delta]^n$ of a submodule \mathcal{M} there exist an efficient method to determine a weighting vector **w** and a column permutation P such that the weak Popov form under $\prec_{\mathbf{w}}$ of $P(\mathcal{M})$ equals the P-permuted *minimal* Gröbner basis of \mathcal{M} under \prec (see [47, Chapter 1.3.4]).

2.2.4 Cost model for skew polynomial operations

For deriving the computational complexity we consider only skew polynomials with zero derivations since in this case explicit results are available in literature. Note, that the isomorphism between $\mathbb{F}_{q^m}[x; \sigma, \delta]$ and $\mathbb{F}_{q^m}[x; \sigma]$ (see Sect. 2.2 and [32, 39]) can be used to obtain complexity bounds for the nonzero derivation case.

We use the big-O notation $O(\cdot)$ to state asymptotic costs of algorithms. Further, we use shorthand $\widetilde{O}(\cdot)$ for $f(n) \in \widetilde{O}(g(n)) \Leftrightarrow \exists k : f(n) \in O(g(n) \log^k g(n))$ which is equivalent to the $O(\cdot)$ notation, ignoring logarithmic factors in the input parameter. We denote by ω the matrix multiplication exponent, i.e. the infimum of values $\omega_0 \in [2; 3]$ such that there is an algorithm for multiplying $n \times n$ matrices over \mathbb{F}_{q^m} in $O(n^{\omega_0})$ operations in \mathbb{F}_{q^m} . The currently best known cost bound in operations in \mathbb{F}_{q^m} is $\omega < 2.37286$ (see [3]).

By $\mathfrak{M}(n)$ we denote the cost of multiplying two skew polynomials from $\mathbb{F}_{q^m}[x; \sigma]$ of degree *n*. The currently best known cost bound for $\mathfrak{M}(n)$ in operations in \mathbb{F}_{q^m} is (see [55])

$$\mathfrak{M}(n) \in O\left(n^{\min\left\{\frac{\omega+1}{2}, 1.635\right\}}\right).$$

🖄 Springer

and (see [16, 17])

$$\mathfrak{M}(n) \in \widetilde{O}\left(\min\left\{n^{\omega-2}m^2, nm^{\omega-1}\right\}\right)$$

in operations in \mathbb{F}_q . Hence, the following skew polynomial operations in $\mathbb{F}_{q^m}[x; \sigma]$ can be performed in $\widetilde{O}(\mathfrak{M}(n))$:

- Left/right division of two skew polynomials of degree at most n
- Generalized operator / remainder evaluation of a skew polynomial of degree at most *n* at *n* elements from \mathbb{F}_{q^m}
- Computation of the minimal polynomials $M_{\mathcal{B}}^{\text{rem}}(x)$ and $M_{\mathcal{B}}^{\text{op}}(x)_{\mathbf{a}}$ for $|\mathcal{B}| \leq n$ w.r.t. the remainder and generalized operator evaluation, respectively
- Computation of the LCLM (see [17, Theorem 3.2.7])

2.3 Skew Kötter–Nielsen–Høholdt interpolation

We now consider the skew KNH interpolation from [33], which is the skew polynomial analogue of the KNH interpolation over ordinary polynomial rings in [63]. Note that due to the isomorphism between $\mathbb{F}_{q^m}[x; \sigma, \delta]$ and the ring of linearized polynomials for σ being the Frobenius automorphism and $\delta = 0$ (zero derivations), the KNH variant over linearized polynomial rings in [65] can be seen as a special case of [33].

As input to our problem, we consider the $n \mathbb{F}_{q^m}$ -linear skew vector evaluation maps¹ \mathscr{E}_i :

$$\mathscr{E}_i: \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1} \to \mathbb{F}_{q^m}$$

where n is the number of interpolation constraints and s is an interpolation parameter.

Later on, we will specify particular mappings \mathscr{E}_i w.r.t. the generalized operator and the remainder evaluation (see Sect. 4). For each skew vector evaluation map \mathscr{E}_i we define the kernels

$$\mathcal{K}_i := \{ \mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1} : \mathscr{E}_i(\mathbf{Q}) = 0 \}, \quad \forall i = 0, \dots, n-1.$$
(33)

For $0 \le i \le n-1$ the intersection $\overline{\mathcal{K}}_i := \mathcal{K}_0 \cap \mathcal{K}_1 \cap \ldots \cap \mathcal{K}_i$ contains all vectors from $\mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1}$ that are mapped to zero under $\mathscr{E}_0, \mathscr{E}_1, \ldots, \mathscr{E}_i$, i.e.

$$\overline{\mathcal{K}}_i = \{ \mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1} : \mathscr{E}_j(\mathbf{Q}) = 0, \forall j = 0, \dots, i \}.$$
(34)

Under the assumption that the $\overline{\mathcal{K}}_i$ are left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodules for all i = 0, ..., n-1 (see [33]) we can state the general skew polynomial vector interpolation problem.

Problem 1 (General Vector Interpolation Problem) Given the integer $s \in \mathbb{Z}_+$, a set of \mathbb{F}_{q^m} -linear vector evaluation maps $\mathcal{E} = \{\mathscr{E}_0, \dots, \mathscr{E}_{n-1}\}$ and a vector $\mathbf{w} \in \mathbb{Z}_+^{s+1}$ compute a wordered weak-Popov Basis for the left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -module

$$\overline{\mathcal{K}}_{n-1} = \{ \mathbf{b} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1} : \mathscr{E}_i(\mathbf{b}) = 0, \forall i = 0, \dots, n-1 \}.$$
(35)

Problem 1 can be solved using a slightly modified variant of the multivariate skew KNH interpolation from [33]. Since the solution of Problem 1 is a w-ordered weak Popov basis for the interpolation module $\overline{\mathcal{K}}_{n-1}$ instead of a single minimal polynomial vector, we modified the output of [33, Algorithm 1] such that it returns a whole basis for the interpolation module

¹ In [33, 63, 65] linear functionals are defined for each interpolation point. Here we give a different definition based on skew polynomial vectors which is equivalent to the definition based on linear functionals if $\mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1}$ is considered as a vector space over \mathbb{F}_{q^m} .

 $\overline{\mathcal{K}}_{n-1}$. A similar approach was used in [8] to construct a basis for the interpolation module over linearized polynomial rings.

The modified multivariate skew KNH interpolation is summarized in Algorithm 1.

Algorithm 1: Modified Skew KNH Interpolation

Input : A set $\{\mathscr{E}_0, \mathscr{E}_1, \dots, \mathscr{E}_{n-1}\}$ of vector evaluation maps A "weighting" vector $\mathbf{w} = (w_0, w_1, \dots, w_s) \in \mathbb{Z}_{\perp}^{s+1}$ **Output**: A w-ordered weak-Popov Basis $\mathbf{B} \in \mathbb{F}_{a^m}[x; \sigma, \delta]^{(s+1)\times(s+1)}$ for $\overline{\mathcal{K}}_{n-1}$ 1 Initialize: $\mathbf{B} = \mathbf{I}_{s+1} \in \mathbb{F}_{a^m}[x; \sigma, \delta]^{(s+1) \times (s+1)}$ **2** for $i \leftarrow 0$ to n - 1 do for $j \leftarrow 0$ to s do 3 4 $\Delta_i \leftarrow \mathscr{E}_i(\mathbf{b}_i)$ $\mathcal{J} \leftarrow \{j : \Delta_i \neq 0\}$ 5 6 if $\mathcal{J} \neq \emptyset$ then $j^* \leftarrow \min_{i \in \mathcal{J}} \{ \arg\min_{i \in \mathcal{J}} \{ \deg_{\mathbf{w}}(\mathbf{b}_i) \} \}$ 7 $\mathbf{b}^* \leftarrow \mathbf{b}_{i^*}$ 8 for $j \in \mathcal{J}$ do 9 if $j = j^*$ then 10 $\mathbf{b}_j \leftarrow \left(x - \frac{\mathscr{E}_i(x\mathbf{b}^*)}{\Delta_{j^*}}\right)\mathbf{b}^*;$ /* degree-increasing step */ 11 12 $\mathbf{b}_j \leftarrow \mathbf{b}_j - \frac{\Delta_j}{\Delta_{j*}} \mathbf{b}^* ;$ /* cross-evaluation step */ 13 14 return B

Note that $\min_{j \in \mathcal{J}} \{ \arg \min_{j \in \mathcal{J}} \{ \deg_{\mathbf{w}}(\mathbf{b}_j) \} \}$ in Line 7 returns the smallest index $j \in \mathcal{J}$ to break ties, i.e. the index j of the minimal vector \mathbf{b}_j w.r.t. $\prec_{\mathbf{w}}$ for which $\Delta_j \neq 0$ (see (32)).

In each iteration of Algorithm 1 (and so [33, Algorithm 1]) there are three possible update steps:

- 1. *No update*: The vector \mathbf{b}_j is not updated if \mathbf{b}_j is in the kernel $\overline{\mathcal{K}}_i$ already, i.e. if $\Delta_j = \mathscr{E}_i(\mathbf{b}_j) = 0$.
- 2. *Cross-evaluation* (or *order-preserving* [33]) update: For any \mathbf{b}_j that is not minimal w.r.t. $\prec_{\mathbf{w}}$ (i.e. $j \neq j^*$) the cross-evaluation update (Line 13) is performed such that

$$\mathscr{E}_i\left(\mathbf{b}_j - \frac{\Delta_j}{\Delta_{j^*}}\mathbf{b}^*\right) = \mathscr{E}_i(\mathbf{b}_j) - \frac{\mathscr{E}_i(\mathbf{b}_j)}{\mathscr{E}_i(\mathbf{b}_{j^*})} \mathscr{E}_i(\mathbf{b}_{j^*}) = 0.$$

Note that the (w-weighted) degree of \mathbf{b}_i is not increased by this update.

3. Degree-increasing (or order-increasing [33]) update: For the minimal vector $\mathbf{b}_{j^*} := \mathbf{b}^*$ w.r.t. $\prec_{\mathbf{w}}$ the degree-increasing update (Line 11) is performed such that

$$\mathscr{E}_{i}\left(\left(x-\frac{\mathscr{E}_{i}(x\mathbf{b}^{*})}{\Delta_{j^{*}}}\right)\mathbf{b}^{*}\right)=\mathscr{E}_{i}(x\mathbf{b}^{*})-\frac{\mathscr{E}_{i}(x\mathbf{b}^{*})}{\mathscr{E}_{i}(\mathbf{b}^{*})}\mathscr{E}_{i}(\mathbf{b}^{*})=0.$$

The (w-weighted) degree of \mathbf{b}^* is increased by one in this case.

The different update steps are illustrated in [33, Figure 1]. Define the sets

$$\mathcal{S}_j := \{ \mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1} : \operatorname{Ind}_{\mathbf{W}}(\mathbf{Q}) = j \} \cup \{ \mathbf{0} \}$$
(36)

🖉 Springer

and

$$\mathcal{T}_{i,j} := \overline{\mathcal{K}}_i \cap \mathcal{S}_j \tag{37}$$

for all i = 0, ..., n - 1 and j = 0, ..., s. Note that $S_j \cap S_{j'} = \{0\}$ for all $0 \le j, j' \le s$. The following result from [33] is fundamental for proving the correctness of Algorithm 1.

Theorem 1 ([33, Theorem 5]) Assume that $\overline{\mathcal{K}}_i$ are left $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -submodules for all i = 0, ..., n - 1. Then after each iteration i of Algorithm 1, the updated \mathbf{b}_j is a minimum w.r.t. $\prec_{\mathbf{w}}$ in $\mathcal{T}_{i,j} = \overline{\mathcal{K}}_i \cap \mathcal{S}_j$ for all j = 0, ..., s.

In other words, after the *i*-th iteration each \mathbf{b}_j has $\operatorname{Ind}_{\mathbf{w}}(\mathbf{b}) = j$ and the minimal **w**-weighted degree among all vectors in $\mathcal{T}_{i,j}$. Therefore, after the *i*-th iteration, the matrix **B** is a **w**-ordered weak Popov basis for $\overline{\mathcal{K}}_i$.

Lemma 4 (Correctness of Algorithm 1) Algorithm 1 is correct and provides a solution to the general vector interpolation problem in Problem 1.

Proof The update steps of Algorithm 1 and [33, Algorithm 1] are equivalent and therefore we have by [33, Theorem 5] that after the *i*-th iteration each $\mathbf{b}_j \in \overline{\mathcal{K}}_i$. We now will show that after the *i*-th iteration of Algorithm 1 the matrix **B** is a **w**-ordered weak Popov basis for $\overline{\mathcal{K}}_i$. By Theorem 1 ([33, Theorem 5]) each \mathbf{b}_j has the minimal **w**-weighted degree among all polynomials in $\mathcal{T}_{i,j}$, which implies that the **w**-pivot indices of $\mathbf{b}_0, \ldots, \mathbf{b}_s$ are increasing and distinct. Now assume that there exists a vector $\mathbf{p} \in \overline{\mathcal{K}}_i$ that can not be represented by a $\mathbb{F}_{q^m}[x; \sigma, \delta]$ -linear combination of the form

$$\mathbf{p} = \sum_{j=0}^{s} a_j \mathbf{b}_j$$

for some $a_i \in \mathbb{F}_{q^m}[x; \sigma, \delta]$. Then we must have that **p** can be written as

$$\mathbf{p} = \mathbf{r} + \sum_{j=0}^{s} a_j \mathbf{b}_j \tag{38}$$

where $\deg_{\mathbf{w}}(\mathbf{r}) < \min_{j} \{\deg_{\mathbf{w}}(\mathbf{b}_{j})\}$. This contradicts that \mathbf{b}_{j} is a minimum w.r.t. $\prec_{\mathbf{w}}$ in $\mathcal{T}_{i,j}$ since $\operatorname{Ind}_{\mathbf{w}}(\mathbf{r}) \in \{0, \ldots, s\}$. Therefore we conclude that after the *i*-th iteration **B** is a **w**-ordered weak Popov basis for $\overline{\mathcal{K}}_{i}$.

Proposition 1 (Computational Complexity of Algorithm 1) *The complexity of Algorithm 1 is dominated by the complexity of:*

- $\mathcal{O}(sn)$ evaluation maps \mathscr{E}_i applied to a vector from $\mathbb{F}_{q^m}[x; \sigma, \delta]_{< n}^{s+1}$,
- *n* multiplications of a monic degree-1 skew polynomial with a vector from $\mathbb{F}_{q^m}[x; \sigma, \delta]_{\leq n}^{s+1}$ (degree-increasing step),
- $\mathcal{O}(sn)$ multiplications of an element from \mathbb{F}_{q^m} with a vector from $\mathbb{F}_{q^m}[x; \sigma, \delta]_{\leq n}^{s+1}$ (cross-evaluation step).

Proof In each of the *n* iterations we have:

- -s + 2 evaluation maps \mathscr{E}_i applied to a vector from $\mathbb{F}_{q^m}[x; \sigma, \delta]_{< n}^{s+1}$ (Line 4),
- one product of a skew polynomial of degree 1 with a vector from $\mathbb{F}_{q^m}[x; \sigma, \delta]_{\leq n}^{s+1}$ (degree-increasing step in Line 11),
- *s* multiplications of an element from \mathbb{F}_{q^m} with a vector from $\mathbb{F}_{q^m}[x; \sigma, \delta]_{\leq n}^{s+1}$ (cross-evaluation step in Line 14),
- -s + 1 inversions/divisions in \mathbb{F}_{q^m} .

3 Fast Kötter–Nielsen–Høholdt interpolation over skew polynomial rings

In [49] a fast D&C variant of the Kötter interpolation for the Guruswami–Sudan decoder for Reed–Solomon codes was presented. We now use ideas from [49] to speed up the skew KNH interpolation from [33]. The main idea is to sub-divide Problem 1 into a tree of successively smaller problems. Each leaf is identified with a linear functional, and the updates done here are represented as skew polynomial matrices. The inner-nodes of the tree combine updates using matrix multiplication. The entire cost of the algorithm hinges on the fact that at any node of the tree, we need only know the intermediate basis up to its image on the linear functionals of the subtree of that node.

In the following, we describe the general framework for the fast skew KNH interpolation algorithm which we will then discuss in Sect. 4 w.r.t. to particular operator and remainder vector evaluation maps.

The operations performed on the basis **B** in the inner loop of the *i*-th iteration of Algorithm 1 can be represented by the matrix

$$\mathbf{U} = \begin{pmatrix} 1 & -\frac{\Delta_{0}}{\Delta_{j^{*}}} \\ \ddots & \vdots \\ 1 & -\frac{\Delta_{j^{*}-1}}{\Delta_{j^{*}}} \\ \left(x - \frac{\mathscr{E}_{i}(x\mathbf{b}_{j^{*}})}{\Delta_{j^{*}}}\right) \\ -\frac{\Delta_{j^{*}+1}}{\Delta_{j^{*}}} & 1 \\ \vdots \\ -\frac{\Delta_{s}}{\Delta_{j^{*}}} & 1 \end{pmatrix}$$
(39)

such that after the *i*-th iteration we obtain the basis **UB** for $\overline{\mathcal{K}}_i$. Note that if $\Delta_j = 0$ no update on the row **b**_j is performed since $\Delta_j / \Delta_{j^*} = 0$.

3.1 Divide-and-Conquer Skew Kötter interpolation

To describe the following algorithms we introduce some notations. $\mathbf{M}_{[i,j]} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1}$ denotes a polynomial vector that is dependent on the index set $\{i, i + 1, ..., j - 1, j\}$ with $j \ge i$ and $\mathbf{M}_{[i,i]} = \mathbf{M}_i$. The (ordered) set \mathcal{M} is globally available for all algorithms and is defined as

$$\mathcal{M} = (\mathbf{M}_{[0,n-1]}, \mathbf{M}_{[0,\lfloor n/2 \rfloor - 1]}, \mathbf{M}_{\lfloor \lfloor n/2 \rfloor, n-1]}, \dots, \mathbf{M}_0, \mathbf{M}_1, \dots, \mathbf{M}_{n-1}) \subseteq \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1}$$

$$(40)$$

for an integer $n \in \mathbb{Z}_+$. For an ordered set (or tuple) of evaluation maps $\mathcal{E} = (\mathscr{E}_0, \ldots, \mathscr{E}_{n-1})$ we use a similar notation to access an ordered subset of \mathcal{E} as follows

$$\mathcal{E}_{[i,j]} = (\mathscr{E}_i, \dots, \mathscr{E}_j). \tag{41}$$

Depending on the considered interpolation problem, we will later on define the polynomial vectors $\mathbf{M}_{[i,j]}$ to contain minimal polynomials that depend on the interpolation points corresponding to the vector evaluation maps in $\mathcal{E}_{[i,j]}$. In the general interpolation problem (Problem 1) we only consider sets of evaluation maps whereas here we consider *ordered* sets since we need the notation within the D&C algorithm to build up the tree.

In order to describe a general framework for the fast skew KNH interpolation, we need the following assumption. In Sect. 4 we show that this assumption holds for specific coding applications.

Assumption 1 Let $\mathcal{E} = \{\mathscr{E}_0, \ldots, \mathscr{E}_{n-1}\}$ be a set of linear functionals as defined in (33) and let $\mathcal{E}_{[i,j]} = \{\mathscr{E}_i, \ldots, \mathscr{E}_j\} \subseteq \mathcal{E}$. We assume that for all $\mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1}$ and $0 \le i \le j \le n-1$ the skew polynomial vector $\mathbf{M}_{[i,j]} \in \mathcal{M}$ (which contains minimal skew polynomials that depend on $\mathcal{E}_{[i,j]}$) satisfies

$$\mathscr{E}_{l}(\mathbf{Q}) = \mathscr{E}_{l}(\mathbf{Q} \bmod_{\mathbf{r}} \mathbf{M}_{[i,j]}), \quad \forall l = i, \dots, j.$$
(42)

Algorithm 2: SkewInterpolatePoint

Input : A skew vector evaluation map \mathcal{E}_i , $\mathbf{B} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{(s+1) \times (s+1)}$ $\mathbf{d} = (d_0, d_1, \dots, d_s) \in \mathbb{Z}^{s+1}_+$ s.t. $d_j = \deg_{\mathbf{w}}(\mathbf{b}_j)$ for all $j = 0, \dots, s$. **Output:** $\mathbf{T} \in \mathbb{F}_{a^m}[x; \sigma, \delta]^{(s+1) \times (s+1)}$ s.t. the rows of $\hat{\mathbf{B}}$:=**TB** is a **w**-ordered weak-Popov Basis for $\langle \mathbf{B} \rangle \cap \hat{\mathcal{K}}_i$ $\hat{\mathbf{d}} = (\hat{d}_0, \hat{d}_1, \dots, \hat{d}_s) \in \mathbb{Z}_+^{s+1}$ s.t. $\hat{d}_j = \deg_{\mathbf{w}}(\hat{\mathbf{b}}_j)$ for all $j = 0, \dots, s$. $1 \hat{\mathbf{d}} \leftarrow \mathbf{d}$ 2 for $i \leftarrow 0$ to s do $\mathbf{3} \mid \Delta_i \leftarrow \mathscr{E}_i(\mathbf{b}_i)$ 4 $\mathcal{J} \leftarrow \{j : d_i \neq 0\}$ 5 $\mathbf{T} \leftarrow \mathbf{I}_{s+1} \in \mathbb{F}_{a^m}[x; \sigma, \delta]^{(s+1) \times (s+1)}$ 6 if $\mathcal{J} \neq \emptyset$ then $j^* \leftarrow \min_{l \in \mathcal{J}} \{ \arg \min_{l \in \mathcal{J}} \{ d_l \} \}$ 7 $T \leftarrow U$ where U is as in (39) 8 $\hat{d}_{i^*} \leftarrow \hat{d}_{i^*} + 1$ 10 return $(\mathbf{T}, \hat{\mathbf{d}})$

Lemma 5 (Correctness of Algorithm 2) SkewInterpolatePoint in Algorithm 2 is correct.

Proof The columns of **U** except for the j^* -th column correspond to the non-minimal rows of **B**. The cross-evaluation step of the *j*-th non-minimal row of **B** in Line 13 of Algorithm 1 is performed by the entries in the *j*-th row and *j**-th column of **U**. The entry in the *j**-th row and the *j**-th column of **U** corresponds to the degree-increasing step in Line 11 in Algorithm 1. Hence, the algorithm outputs a matrix **T** such that all rows of **TB** are mapped to zero under \mathscr{E}_i .

Equipped with the routine SkewInterpolatePoint in Algorithm 2 to solve the basic step we can now derive a D&C variant of the skew KNH interpolation in Algorithm 1.

Lemma 6 (Correctness of Algorithm 3) Under Assumption 1, SkewInterpolateTree in Algorithm 3 is correct.

Proof The correctness of Algorithm 3 follows directly from the definition of the matrix U in (39) and Assumption 1.

Algorithm 3: SkewInterpolateTree

Input : Skew vector evaluation maps $\mathcal{E}_{[i_1,i_2]} = (\mathscr{E}_{i_1}, \dots, \mathscr{E}_{i_2})$ $\mathbf{B} \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{(s+1) \times (s+1)},$ $\mathbf{d} = (d_0, d_1, \dots, d_s) \in \mathbb{Z}^{s+1}_+$ s.t. $d_j = \deg_{\mathbf{w}}(\mathbf{b}_j)$ for all $j = 0, \dots, s$. **Output:** A matrix $\mathbf{T} \in \mathbb{F}_{a^m}[x; \sigma, \delta]^{(s+1) \times (s+1)}$ s.t. $\hat{\mathbf{B}} := \mathbf{TB}$ is a w-ordered weak-Popov Basis for $\langle \mathbf{B} \rangle \cap \mathcal{K}_{i_1} \cap \cdots \cap \mathcal{K}_{i_2},$ $\hat{\mathbf{d}} = (\hat{d}_0, \hat{d}_1, \dots, \hat{d}_s) \in \mathbb{Z}^{s+1}_+$ s.t. $\hat{d}_j = \deg_{\mathbf{w}}(\hat{\mathbf{b}}_j)$ for all $j = 0, \dots, s$. 1 **if** $i_1 = i_2$ **then** 2 **return** SkewInterpolatePoint(\mathcal{E}_{i_1} , **B**, **d**) 3 else $z \leftarrow \left\lfloor \frac{i_1 + i_2}{2} \right\rfloor$ $\mathbf{B}_1 \leftarrow \mathbf{B} \bmod_r \mathbf{M}_{[i_1, z]}$ 4 5 $(\mathbf{T}_1, \mathbf{d}_1) \leftarrow \mathsf{SkewInterpolateTree}(\mathcal{E}_{[i_1, z]}, \mathbf{B}_1, \mathbf{d})$ 6 $\mathbf{B}_2 \leftarrow \mathbf{T}_1 \mathbf{B} \mod_r \mathbf{M}_{[z+1,i_2]}$ 7 8 $(\mathbf{T}_2, \mathbf{d}_2) \leftarrow \mathsf{SkewInterpolateTree}(\mathcal{E}_{[z+1, i_2]}, \mathbf{B}_2, \mathbf{d}_1)$ return ($\mathbf{T} = \mathbf{T}_2 \mathbf{T}_1$, $\hat{\mathbf{d}} = \mathbf{d}_2$) 9

3.2 Precomputing minimal polynomial vectors

We now present a generic procedure to pre-compute the set \mathcal{M} containing the minimal polynomial vectors $\mathbf{M}_{[i,j]}$ required in Algorithm 3 efficiently. We consider minimal polynomials such as the generalized operator and the remainder evaluation which can be constructed by means of the LCLM of polynomial sequences (see (14) and (23)) we use the ideas from [17, Theorem 3.2.7] to obtain the efficient procedure described in Algorithm 4. The D&C structure of the algorithm is illustrated in Fig. 1 for an example of $\kappa = 4$. The initial minimal polynomial vectors $\mathbf{M}_0, \mathbf{M}_1, \ldots, \mathbf{M}_{\kappa-1}$ from which all other minimal polynomials are computed via the LCLM, are computed depending on the application. In Sect. 4 two cases are given, for the general operator evaluation as in (55) and for the remainder evaluation as in (86).

Algorithm 4: PreComputeMinVectorsTree

Input : Upper and lower index bound $a \in \mathbb{Z}_+$ and $b \in \mathbb{Z}_+$ with $b \ge a$ Minimal polynomial vectors $\mathbf{M}_a, \mathbf{M}_{a+1}, \dots, \mathbf{M}_b \in \mathbb{F}_{q^m}[x; \sigma, \delta]^{s+1}$ Output: An (ordered) set $(\mathbf{M}_{[a,b]},\mathbf{M}_{[a,\lfloor(b-1)/2\rfloor-1]},\mathbf{M}_{[\lfloor(b-1)/2\rfloor,b]},\ldots,\mathbf{M}_{[a,a]},\mathbf{M}_{[a+1,a+1]},\ldots,\mathbf{M}_{[b,b]})\subset$ $\mathbb{F}_{a^m}[x;\sigma,\delta]^{s+1}$ 1 if a = b then 2 return {M_a} 3 else $\delta \leftarrow \lfloor \frac{b-a+1}{2} \rfloor$ 4 $\mathcal{M}_1 \leftarrow \operatorname{PreComputeMinVectorsTree}(a, a + \delta - 1)$ 5 $\mathcal{M}_2 \leftarrow \operatorname{PreComputeMinVectorsTree}(a + \delta, b)$ 6 $\mathbf{M}_{[a,b]} \leftarrow \operatorname{lclm}\left(\mathbf{M}_{[a,a+\delta-1]},\mathbf{M}_{[a+\delta,b]}\right); \text{ /* with } \mathbf{M}_{[a,a+\delta-1]} \in \mathcal{M}_1 \text{ and } \mathbf{M}_{[a+\delta,b]} \in \mathcal{M}_2$ 7 * / return $\mathcal{M}_1 \cup \mathcal{M}_2 \cup \{\mathbf{M}_{[a,b]}\}$ 8



Fig. 1 Illustration of the computation tree of Algorithm 4 to precompute all minimal polynomial vectors in the set M for $\kappa = 4$

Lemma 7 (Correctness of Algorithm 4) Algorithm 4 is correct.

Proof The correctness of Algorithm 4 follows directly from [17, Theorem 3.2.7]. The algorithm proceeds in a recursive manner and splits the size of the set of considered minimal polynomials in half. When sets consist only of one element, $\mathbf{M}_{[a,a]}$ are computed, using the generalized operator or remainder evaluation (see (14) and (23)). The sets of minimal polynomials of larger size are then obtained by merging the smaller sets of minimal polynomials using the relation $\mathbf{M}_{[a,b]} = \text{lclm} \left(\mathbf{M}_{[a,a+\delta-1]}, \mathbf{M}_{[a+\delta,b]} \right)$ with $\delta = \lfloor \frac{b-a+1}{2} \rfloor$, also illustrated in Fig. 1.

4 Application to coding problems

In this section we apply the fast skew KNH interpolation described in Sect. 3 to coding problems in the rank, sum-rank and skew metric. For simplicity, we consider evaluation codes constructed over $\mathbb{F}_{q^m}[x; \sigma]$, i.e. skew polynomial rings with zero derivations, only. The results can be generalized to the $\mathbb{F}_{q^m}[x; \sigma, \delta]$ case where the complexity analysis can be performed using the ring isomorphism to $\mathbb{F}_{q^m}[x; \sigma]$ (see Sect. 2.2).

First we consider the application to interpolation-based decoding of interleaved Gabidulin codes in the rank metric. As interleaved Gabidulin codes are a special case of interleaved linearized Reed–Solomon (ILRS) codes, the decoder derived in Sect. 4.1 is a special case of the decoder for interleaved linearized Reed–Solomon (LRS) codes derived in Sect. 4.2. Nevertheless, we state the application for decoding interleaved Gabidulin codes in the rank metric separately to make it more accessible to readers that are familiar only with the rank metric.

In Sect. 4.3 we show how to apply the fast KNH interpolation algorithm to decoding interleaved skew Reed–Solomon (ISRS) codes in the skew metric. In contrast to the decoding approaches the rank and the sum-rank metric, the decoder performs the interpolation step with respect to the remainder evaluation and thus does not require the isometry between the sum-rank and the skew metric from [39].

4.1 Interpolation-based decoding of interleaved Gabidulin codes

Interleaved Gabidulin codes are rank-metric codes that are obtained by the Cartesian product of ordinary Gabidulin codes [20] and which allow for decoding beyond half the minimum

rank distance (see [35, 54]). In [35], a probabilistic unique decoder that is able to correct errors beyond the unique decoding radius with high probability, was presented. A Berlekamp–Massey-like decoding algorithm for interleaved Gabidulin codes was presented in [57, 59].

A Welch–Berlekamp-like interpolation-based decoding scheme, that can be either used as a (not necessarily polynomial-time) list decoding algorithm or as a probabilistic unique decoding algorithm was presented by Wachter-Zeh and Zeh in [62]. The algorithm consists of an *interpolation step* and a *root-finding step*. The list size (and so the probability of obtaining a unique solution) can be optimized by using a minimal Gröbner basis for the left module containing the solutions of the interpolation problem rather than a single solution of the interpolation problem for the root-finding step (see [8, 9]).

The interpolation step (including the computation of the minimal Gröbner basis) can be solved by the linearized or skew variants of the multivariate KNH interpolation [33, 65] in $O(s^2n^2)$ operations in \mathbb{F}_{q^m} , where *s* is the *interleaving order* and *n* is the length of the code. The overall computational complexity of the algorithm in [62] is then in the order of $O(s^2n^2)$ operations in \mathbb{F}_{q^m} .

Recently, efficient algorithms for solving the interpolation step and the root-finding step that are based on the computation of minimal approximant bases over skew polynomial rings were presented in [11]. The algorithms reduce the overall decoding complexity of the Wachter-Zeh and Zeh decoder [62] to $\tilde{O}(s^{\omega}\mathfrak{M}(n)) \subseteq \tilde{O}(s^{\omega}n^{1.635})$.

In this section we show how the fast skew KNH interpolation algorithm from Sect. 3 can be used to accomplish the interpolation step in the Wachter-Zeh and Zeh decoder in $\widetilde{O}(s^{\omega}\mathfrak{M}(n)) \subseteq \widetilde{O}(s^{\omega}n^{1.635})$ operations in \mathbb{F}_{q^m} . The obtained improved computational complexity coincides with the computational complexity of the minimal approximant bases variant from [11]. The results can be also used to speed up the interpolation-based decoder for interleaved subspace codes in [9].

4.1.1 Codes in the rank metric

The *rank weight* of a vector $\mathbf{x} \in \mathbb{F}_{a^m}^n$ is defined as (see [20])

$$wt_R(\mathbf{x}) := rk_q(\mathbf{x}) = rk_q(\mathbf{X})$$
(43)

where $\mathbf{X} \in \mathbb{F}_q^{m \times n}$ is the corresponding expanded matrix of \mathbf{x} over \mathbb{F}_q . Notice, that in general we have that $\operatorname{wt}_R(\mathbf{x}) \leq \operatorname{wt}_H(\mathbf{x})$ for any $\mathbf{x} \in \mathbb{F}_{q^m}^n$ (see [38]). The *rank distance* between two vectors $\mathbf{x}, \mathbf{y} \in \mathbb{F}_{q^m}^n$ is then defined as the rank of their difference, i.e. as

$$d_R(\mathbf{x}, \mathbf{y}) := \operatorname{wt}_R(\mathbf{x} - \mathbf{y}). \tag{44}$$

The *rank weight* of a matrix $\mathbf{X} \in \mathbb{F}_{q^m}^{s \times n}$ is defined as the \mathbb{F}_q -rank of the matrix $\mathbf{X}_q \in \mathbb{F}_q^{sm \times n}$ that is obtained by expanding each row of \mathbf{X} with respect to a fixed basis of \mathbb{F}_{q^m} over \mathbb{F}_q . Note that the \mathbb{F}_q -rank wt_R (\mathbf{X}) can be equivalently seen as the maximum number of \mathbb{F}_q -linearly independent columns of \mathbf{X} .

As a channel model we consider the rank error channel

$$\mathbf{R} = \mathbf{C} + \mathbf{E} \tag{45}$$

where the error matrix **E** is chosen uniformly at random from all matrices in $\mathbb{F}_{q^m}^{s \times n}$ with $\operatorname{wt}_R(\mathbf{E}) = t$.

Definition 7 (Interleaved Gabidulin Code [35, 54]) Let σ be the Frobenius automorphism $\sigma(\cdot) = \cdot^q$ of \mathbb{F}_{q^m} . Let $\boldsymbol{\beta} = (\beta_0, \beta_1, \dots, \beta_{n-1}) \in \mathbb{F}_{q^m}^n$ contain \mathbb{F}_q -linearly independent

elements from \mathbb{F}_{q^m} . An *s*-interleaved Gabidulin code of length $n \leq m$ and dimension k is defined as

$$\operatorname{IGab}[\boldsymbol{\beta}, s; n, k] = \left\{ \begin{pmatrix} f^{(1)}(\boldsymbol{\beta})_1 \\ f^{(2)}(\boldsymbol{\beta})_1 \\ \vdots \\ f^{(s)}(\boldsymbol{\beta})_1 \end{pmatrix} : f^{(j)} \in \mathbb{F}_{q^m}[x; \sigma]_{< k}, \forall j \in [1, s] \right\}.$$
(46)

Interleaved Gabidulin codes fulfill the Singleton-like bound in the rank metric with equality, i.e. we have that $d_R(\text{IGab}[\boldsymbol{\beta}, s; n, k]) = n - k + 1$, and thus are maximum rank distance (MRD) codes.

Suppose we transmit a codeword

$$\mathbf{C} = \begin{pmatrix} f^{(1)}(\boldsymbol{\beta})_1 \\ f^{(2)}(\boldsymbol{\beta})_1 \\ \vdots \\ f^{(s)}(\boldsymbol{\beta})_1 \end{pmatrix} \in \mathrm{IGab}[\boldsymbol{\beta}, s; n, k]$$
(47)

and receive

$$\mathbf{R} = \mathbf{C} + \mathbf{E} \tag{48}$$

where $\operatorname{wt}_R(\mathbf{E}) = t$.

Definition 8 (*Generalized Operator Vector Evaluation Map*) Given an interpolation point set $\mathcal{P} = \{(p_{i,0}, p_{i,1}, \dots, p_{i,s}) : i = 0, \dots, n-1\} \subseteq \mathbb{F}_{q^m}^{s+1}$, a vector $\mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$, and a vector $\mathbf{a} \in \mathbb{F}_{q^m}^n$ containing the generalized operator evaluation parameters, we define the generalized vector evaluation maps as

$$\mathscr{E}_{i}^{op}(\mathbf{Q})_{a_{i}} := \sum_{j=0}^{s} Q_{j}(p_{i,j})_{a_{i}}, \quad \forall i = 0, \dots, n-1.$$
(49)

For interleaved Gabidulin codes, the interpolation point set is

$$\mathcal{P} = \{ (\beta_i, r_{0,i}, \dots, r_{s-1,i}) : i = 0, \dots, n-1 \} \subset \mathbb{F}_{q^m}^{s+1}$$
(50)

with $\mathbf{a} = (1, 1, ..., 1) \in \mathbb{F}_{q^m}^n$ being a vector containing the corresponding generalized operator evaluation parameters. Equipped with these definitions the interpolation problem [62, Problem 1] for decoding interleaved Gabidulin codes can be stated as follows.

Problem 2 (Vector Interpolation Problem) Given the integer $s \in \mathbb{Z}_+$, a degree constraint $D \in \mathbb{Z}_+$, a set of \mathbb{F}_{q^m} -linear vector evaluation maps $\mathcal{E}^{op} = \{\mathcal{E}_0^{op}, \ldots, \mathcal{E}_{n-1}^{op}\}$ as defined in (49), the vector $\mathbf{a} = (1, 1, \ldots, 1) \in \mathbb{F}_{q^m}^n$ and a vector $\mathbf{w} = (0, k - 1, \ldots, k - 1) \in \mathbb{Z}_+^{s+1}$ compute a vector $\mathbf{b} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ that satisfies:

 $-\mathscr{E}_i^{op}(\mathbf{b})_1 = 0, \forall i = 0, \dots, n-1,$ $- \deg_{\mathbf{w}}(\mathbf{b}) < D.$

A nonzero solution of Problem 2 exists if the degree constraint satisfies (see [62])

$$D = \left\lceil \frac{n+s(k-1)+1}{s+1} \right\rceil.$$
(51)

Proposition 2 Problem 2 can be solved by Algorithm 1 in $O(s^2n^2)$ operations in \mathbb{F}_{q^m} .

Proof Problem 2 corresponds to finding a vector in the interpolation module

$$\overline{\mathcal{K}}_{n-1} = \{ \mathbf{b} \in \mathbb{F}_{q^m}[x;\sigma]^{s+1} : \mathscr{E}_i^{op}(\mathbf{b})_1 = 0, \forall i = 0, \dots, n-1 \}$$
(52)

of minimal **w**-weighted degree. Hence, Problem 2 can be solved by computing a **w**-ordered weak-Popov basis for the left $\mathbb{F}_{q^m}[x; \sigma]$ -module $\overline{\mathcal{K}}_{n-1}$ and returning a row with **w**-weighted degree less than *D*. The problem of finding a **w**-ordered weak-Popov basis for $\overline{\mathcal{K}}_{n-1}$ is an instance of the general vector interpolation problem (Problem 1) which can be solved by Algorithm 1 (see Lemma 4). By Proposition 1, Algorithm 1 requires *O*(*sn*) evaluation maps \mathcal{E}_i^{op} of a vector in $\mathbb{F}_{q^m}[x; \sigma]_{\leq n}^{s+1}$, which requires *O*(*sn*) operations in \mathbb{F}_{q^m} each. Overall, the computation of the evaluation maps requires *O*(*sn*) operations in \mathbb{F}_{q^m} . The computation of the evaluation maps requires *O*(*s*²*n*²) operations in \mathbb{F}_{q^m} . The computations of a monic degree-1 skew polynomial with a vector from $\mathbb{F}_{q^m}[x; \sigma]$ (Line 11) requires *O*(*sn*²) operations in \mathbb{F}_{q^m} in total. The *O*(*sn*) multiplications of an element from \mathbb{F}_{q^m} and a vector from $\mathbb{F}_{q^m}[x; \sigma]_{\leq n}^{s+1}$ require at most *O*(*s*²*n*²) operations in \mathbb{F}_{q^m} .

If the error weight $t = \operatorname{wt}_R(\mathbf{E})$ satisfies

$$t < \frac{s}{s+1}(n-k+1)$$
(53)

it can be shown that (see [62, Theorem 1])

$$P(x) := Q_0(x) + \sum_{j=1}^{s} Q_j(x) f^{(j)}(x) = 0.$$
(54)

The root-finding step consists of finding all polynomials $f^{(1)}, \ldots, f^{(s)} \in \mathbb{F}_{q^m}[x; \sigma]_{< k}$ that satisfy (54). This task can be accomplished by the minimal approximant basis methods in [10, 11] requiring at most \widetilde{O} ($s^{\omega}\mathfrak{M}(n)$) operations in \mathbb{F}_{q^m} .

4.1.2 Solving the interpolation step via the fast KNH interpolation

For an interpolation point set $\mathcal{P} = {\mathbf{p}_0, \mathbf{p}_1, \dots, \mathbf{p}_{n-1}} \subseteq \mathbb{F}_{q^m}^{s+1}$ define the vectors of minimal polynomials with respect to the generalized operator evaluation for $0 \le i, j \le n-1$ and $i \le j$ as

$$\mathbf{M}_{[i,j]}^{\text{op}}(x)_{a_i} := \left(M_{\{p_{i,0},\dots,p_{j,0}\}}^{\text{op}}(x)_{a_i}, M_{\{p_{i,1},\dots,p_{j,1}\}}^{\text{op}}(x)_{a_i},\dots, M_{\{p_{i,s},\dots,p_{j,s}\}}^{\text{op}}(x)_{a_i} \right).$$
(55)

Lemma 8 Let $\mathcal{E}^{op} = (\mathscr{E}_0^{op}, \dots, \mathscr{E}_{n-1}^{op})$ be an ordered set of skew vector evaluation maps as defined in (49) and let $\mathcal{E}_{[i,j]}^{op} = (\mathscr{E}_i^{op}, \dots, \mathscr{E}_j^{op}) \subseteq \mathcal{E}^{op}$. Then for any $\mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ we have that

$$\mathscr{E}_{l}^{op}(\mathbf{Q})_{a_{l}} = \mathscr{E}_{l}^{op}(\mathbf{Q} \bmod_{\mathbf{r}} \mathbf{M}_{[i,j]}^{op}(x)_{\mathbf{a}})_{a_{l}}, \quad \forall l = i, \dots, j$$
(56)

where $\mathbf{a} = (a_i, \ldots, a_j)$ contains the corresponding general operator evaluation parameters.

Proof The lemma follows directly by applying the result from Lemma 1 to the elementary evaluations in the skew vector operator evaluation maps defined in (49). \Box

An important consequence of Lemma 8 is, that the generalized operator vector evaluation maps from Definition 8 and the minimal polynomial vectors in (55) fulfill Assumption 1. Hence, we can solve Problem 2 by calling Algorithm 3 with \mathcal{E}^{op} , basis \mathbf{I}_{s+1} and initial degrees $(0, k - 1, \dots, k - 1)$.

4.1.3 Complexity analysis

We now perform a complexity analysis of Algorithm 4 for the generalized operator evaluation maps defined in (49) ($\mathbb{F}_{q^m}[x; \sigma]$ -case).

Lemma 9 (Complexity of Computing Minimal Polynomial Vectors) Algorithm 4 constructs the (ordered) set $\mathcal{M}^{op} \subset \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ containing the minimal polynomial vectors defined as

$$\mathcal{M}^{op} := \left(\mathbf{M}^{op}_{[0,n-1]}(x)_{\mathbf{a}}, \mathbf{M}^{op}_{[0,\lfloor n/2 \rfloor - 1]}(x)_{\mathbf{a}}, \mathbf{M}^{op}_{\lfloor \lfloor n/2 \rfloor, n-1]}(x)_{\mathbf{a}}, \dots, \mathbf{M}^{op}_{\lfloor n-1, n-1]}(x)_{\mathbf{a}} \right)$$
(57)

in $\widetilde{O}(s\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

Proof Algorithm 4 is a generalization of the procedure in [17, Theorem 3.2.7] to construct a single minimal polynomial, which requires $\tilde{O}(\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} . Hence, the overall complexity of Algorithm 4 is in the order of $\tilde{O}(s\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

Theorem 2 (Computational Complexity) Algorithm 3 solves Problem 2 over $\mathbb{F}_{q^m}[x; \sigma]^{s+1}$ in $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

Proof Let C(n) denote the complexity on n input points without the cost of the routine SkewInterpolatePoint (see Algorithm 2). We have that deg(\mathbf{T}_1), deg(\mathbf{T}_2) $\leq n/2$ and thus the product of $\mathbf{T}_2\mathbf{T}_1$ in Line 9 requires $\widetilde{O}(s^{\omega}\mathfrak{M}(n/2)) \in \widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} . By the master theorem we have $C(n) = 2C(n/2) + \widetilde{O}(s^{\omega}n/2)$ implying that $C(n) \in \widetilde{O}(s^{\omega}n)$. The complexity of SkewInterpolatePoint is dominated by s + 1 univariate skew polynomials of degree less than 1, which requires $O(s^2)$ operations in \mathbb{F}_{q^m} .

The routine SkewInterpolatePoint is called *n* times yielding $O(s^2n)$ operations in total. By Lemma 9 all minimal polynomial vectors required in Lines 5 and 7 of Algorithm 3 can be pre-computed in $\widetilde{O}(s\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} . One (right) modulo operation in $\mathbb{F}_{q^m}[x; \sigma]^{s+1}$ requires $O(s\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} . Therefore, Lines 5 and 7 require $\widetilde{O}(s^2\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} each.

Remark 1 (Practical Consideration) Note that in the $\mathbb{F}_{q^m}[x; \sigma]$ -case the *i*-th generalized operator vector evaluation map $\mathscr{E}_i^{op}(x\mathbf{Q})_{a_i}$ of $x\mathbf{Q}$ can be computed efficiently from $\mathscr{E}_i^{op}(\mathbf{Q})_{a_i}$ requiring only one application of the automorphism σ and one multiplication since by the product rule [40] we have that

$$\mathscr{E}_{i}^{op}(\mathbf{x}\mathbf{Q})_{a_{i}} = \sigma\left(\mathscr{E}_{i}^{op}(\mathbf{Q})_{a_{i}}\right)a_{i}, \quad \forall i = 0, \dots, n-1.$$
(58)

4.2 Interpolation-based decoding of interleaved linearized Reed–Solomon codes

Linearized Reed–Solomon (LRS) codes are codes that have distance properties with respect to the *sum-rank* metric and were introduced in [39] and also considered in [15]. Recently, codes in the sum-rank metric gained attraction since they generalize several code families in the Hamming metric and the rank metric, such as Reed–Solomon codes and Gabidulin codes, and have potential applications in code-based quantum-resistant cryptosystems [56]. Further applications include the construction of space-time codes [36], locally repairable codes with maximal recoverability [44] (also known as partial MDS codes) and error control for multishot network coding [43]. Interpolation-based decoding of *s*-interleaved LRS codes, which allows for correcting errors beyond the unique decoding radius (up to $\frac{s}{s+1}(n-k+1)$) in the sum-rank metric, was recently considered in [6].

In the following we show, how Algorithm 3 can be used to solve the interpolation step in the interpolation-based decoder for ILRS from [6] efficiently.

The sum-rank weight of a vector $\mathbf{x} = (\mathbf{x}^{(1)} | \mathbf{x}^{(2)} | \cdots | \mathbf{x}^{(\ell)}) \in \mathbb{F}_{q^m}^n$, where $\mathbf{x}^{(\lambda)} \in \mathbb{F}_{q^m}^{n_\lambda}$ for all $\lambda = 1, \ldots, \ell$, is defined as (see [36, 51])

$$\operatorname{wt}_{\Sigma R}(\mathbf{x}) := \sum_{\lambda=1}^{\ell} \operatorname{rk}_{q}\left(\mathbf{x}^{(\lambda)}\right).$$
(59)

For any $\mathbf{x} \in \mathbb{F}_{q^m}^n$ we have that $\operatorname{wt}_{\Sigma R}(\mathbf{x}) \leq \operatorname{wt}_H(\mathbf{x})$ (see [38, 39]). The sum-rank distance between two vectors $\mathbf{x}, \mathbf{y} \in \mathbb{F}_{q^m}^n$ is then

$$d_{\Sigma R}(\mathbf{x}, \mathbf{y}) := \operatorname{wt}_{\Sigma R}(\mathbf{x} - \mathbf{y}) = \sum_{i=\lambda}^{\ell} \operatorname{rk}_{q}\left(\mathbf{x}^{(\lambda)} - \mathbf{y}^{(\lambda)}\right).$$
(60)

The sum-rank weight of a matrix $\mathbf{X} = (\mathbf{X}^{(1)} | \mathbf{X}^{(2)} | \cdots | \mathbf{X}^{(\ell)}) \in \mathbb{F}_{q^m}^{s \times n}$ is defined as

$$\operatorname{wt}_{\Sigma R}(\mathbf{X}) := \sum_{\lambda=1}^{\ell} \operatorname{rk}_{q}\left(\mathbf{X}^{(\lambda)}\right), \tag{61}$$

where $\mathbf{X}^{(\lambda)} \in \mathbb{F}_{q^m}^{s \times n_{\lambda}}$ for all $\lambda = 1, \dots, \ell$. The sum-rank distance between two matrices $\mathbf{X}, \mathbf{Y} \in \mathbb{F}_{q^m}^{s \times n}$ is then defined as

$$d_{\Sigma R}(\mathbf{X}, \mathbf{Y}) := \operatorname{wt}_{\Sigma R}(\mathbf{X} - \mathbf{Y}) = \sum_{\lambda=1}^{\ell} \operatorname{rk}_{q}\left(\mathbf{X}^{(\lambda)} - \mathbf{Y}^{(\lambda)}\right).$$
(62)

As channel model we consider the sum-rank channel

$$\mathbf{R} = \mathbf{C} + \mathbf{E} \tag{63}$$

where the error matrix

$$\mathbf{E} = (\mathbf{E}^{(1)} \mid \mathbf{E}^{(2)} \mid \dots \mid \mathbf{E}^{(\ell)})$$
(64)

is chosen uniformly at random from all matrices in $\mathbb{F}_{q^m}^{s \times n}$ with sum-rank weight wt_{ΣR}(\mathbf{e}) = t.

Definition 9 (Interleaved Linearized Reed–Solomon Code) Let σ be the Frobenius automorphism of \mathbb{F}_{q^m} defined as $\sigma(\cdot) = \cdot^q$. Let $\boldsymbol{\xi} = (\xi_0, \xi_1, \dots, \xi_{\ell-1}) \in \mathbb{F}_{q^m}^{\ell}$ be a vector containing representatives from different conjugacy classes of \mathbb{F}_{q^m} . Let the vectors $\boldsymbol{\beta}^{(\lambda)} = \left(\beta_0^{(\lambda)}, \dots, \beta_{n_{\lambda}-1}^{(\lambda)}\right) \in \mathbb{F}_{q^m}^{n_{\lambda}}$ contain \mathbb{F}_q -linearly independent elements from \mathbb{F}_{q^m} for all $\lambda = 1, \dots, \ell$ and define the vector $\boldsymbol{\beta} = \left(\boldsymbol{\beta}^{(1)}, \boldsymbol{\beta}^{(2)}, \dots, \boldsymbol{\beta}^{(\ell)}\right) \in \mathbb{F}_{q^m}^n$. An *s*-interleaved linearized Reed–

Solomon (ILRS) code of length $n = n_1 + n_2 + \cdots + n_\ell$ and dimension k is defined as

$$ILRS[\boldsymbol{\beta}, \ell, s; n, k] = \left\{ \begin{pmatrix} f^{(1)}(\boldsymbol{\beta}^{(1)})_{\xi_0} & \dots & f^{(1)}(\boldsymbol{\beta}^{(\ell)})_{\xi_{\ell-1}} \\ f^{(2)}(\boldsymbol{\beta}^{(1)})_{\xi_0} & \dots & f^{(2)}(\boldsymbol{\beta}^{(\ell)})_{\xi_{\ell-1}} \\ \vdots & \ddots & \vdots \\ f^{(s)}(\boldsymbol{\beta}^{(1)})_{\xi_0} & \dots & f^{(s)}(\boldsymbol{\beta}^{(\ell)})_{\xi_{\ell-1}} \end{pmatrix} : \begin{array}{c} f^{(j)} \in \mathbb{F}_{q^m}[x; \sigma]_{< k}, \\ \forall j \in [1, s] \\ \forall j \in [1, s] \end{array} \right\}.$$
(65)

The code rate and the minimum sum-rank distance of ILRS codes is $R = \frac{k}{n}$ and d =n-k+1 (see [6]), respectively. ILRS codes fulfill the Singleton-like bound in the sum-rank metric with equality and thus are maximum sum-rank distance (MSRD) codes [39].

Suppose we transmit a codeword

$$\mathbf{C} = \begin{pmatrix} f^{(1)}(\boldsymbol{\beta}^{(1)})_{\xi_{0}} & \dots & f^{(1)}(\boldsymbol{\beta}^{(\ell)})_{\xi_{\ell-1}} \\ f^{(2)}(\boldsymbol{\beta}^{(1)})_{\xi_{0}} & \dots & f^{(2)}(\boldsymbol{\beta}^{(\ell)})_{\xi_{\ell-1}} \\ \vdots & \ddots & \vdots \\ f^{(s)}(\boldsymbol{\beta}^{(1)})_{\xi_{0}} & \dots & f^{(s)}(\boldsymbol{\beta}^{(\ell)})_{\xi_{\ell-1}} \end{pmatrix} \in \mathrm{ILRS}[\boldsymbol{\beta}, \ell, s; n, k]$$
(66)

and receive a matrix

$$\mathbf{R} = \left(\mathbf{R}^{(1)} \mid \mathbf{R}^{(2)} \mid \dots \mid \mathbf{R}^{(\ell)}\right) = \mathbf{C} + \mathbf{E}$$
(67)

with $\mathbf{R}^{(\lambda)} \in \mathbb{F}_{q^m}^{s \times n_{\lambda}}$ for all $\lambda = 1, ..., \ell$ where the error matrix $\mathbf{E} = (\mathbf{E}^{(1)} | \mathbf{E}^{(2)} | \cdots | \mathbf{E}^{(\ell)})$ has sum-rank weight $wt_{\Sigma R}(\mathbf{E}) = t$. In order to simplify the notation we index the entries in **R** as $r_{i,i}$ for j = 0, ..., s - 1 and i = 0, ..., n - 1.

Further, we define the vector

$$\mathbf{a} = (\mathbf{a}^{(1)}, \mathbf{a}^{(2)}, \dots, \mathbf{a}^{(\ell)}) \in \mathbb{F}_{q^m}^n$$
(68)

where $\mathbf{a}^{(\lambda)} = (\xi_{\lambda-1}, \dots, \xi_{\lambda-1}) \in \mathbb{F}_{q^m}^{n_\lambda}$ for all $\lambda = 1, \dots, \ell$. For interleaved linearized Reed–Solomon codes, the interpolation point set is

$$\mathcal{P} = \{ (\beta_i, r_{0,i}, \dots, r_{s-1,i}) : i = 0, \dots, n-1 \} \subset \mathbb{F}_{q^m}^{s+1}.$$
(69)

Problem 3 (Vector Interpolation Problem) Given the integer $s \in \mathbb{Z}_+$, a degree constraint $D \in \mathbb{Z}_+$, a set of \mathbb{F}_{q^m} -linear vector evaluation maps $\mathcal{E}^{op} = \{\mathcal{E}_0^{op}, \dots, \mathcal{E}_{n-1}^{op}\}$ as defined in (49), a vector $\mathbf{a} = (a_0, a_1, \dots, a_{n-1}) \in \mathbb{F}_{a^m}^n$ as defined in (68) and a vector $\mathbf{w} = (0, k - 1)$ $1, \ldots, k-1 \in \mathbb{Z}^{s+1}_+$ compute a vector $\mathbf{b} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ that satisfies:

 $-\mathscr{E}_i^{op}(\mathbf{b})_{a_i}=0,\,\forall i=0,\ldots,n-1,$ $- \deg_{\mathbf{w}}(\mathbf{b}) < D.$

A nonzero solution of Problem 3 exists if the degree constraint satisfies

$$D = \left\lceil \frac{n+s(k-1)+1}{s+1} \right\rceil.$$
(70)

By following the ideas in Proposition 2 for general evaluation parameters we see that Problem 3 can be solved by Algorithm 1 in $O(s^2n^2)$ operations in \mathbb{F}_{q^m} .

If the sum-rank weight of the error $t = \operatorname{wt}_{\Sigma R}(\mathbf{E})$ satisfies

$$t < \frac{s}{s+1}(n-k+1)$$
(71)

Springer

we have that (see [6, Theorem 2])

$$P(x) = Q_0(x) + \sum_{j=1}^{s} Q_j(x) f^{(j)}(x) = 0.$$
(72)

The root-finding problem consists of finding all polynomials $f^{(1)}, \ldots, f^{(s)} \in \mathbb{F}_{q^m}[x; \sigma]_{<k}$ that satisfy (72). The root-finding problem can be solved efficiently by the minimal approximant basis methods in [10, 11] requiring at most $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

4.2.1 Solving the interpolation step via the fast KNH interpolation

Since Lemma 8 holds for arbitrary evaluation parameters **a** we have that the minimal polynomial vectors in (55) fulfill Assumption 1. Consequently, we can solve Problem 3 by calling Algorithm 3 with \mathcal{E}^{op} as defined in (49), the basis \mathbf{I}_{s+1} and initial degrees $(0, k-1, \ldots, k-1)$.

4.2.2 Complexity analysis

Note that Problem 2 is an instance of Problem 3 with particular evaluation parameters. Therefore, the complexity follows directly from Theorem 2.

Corollary 1 (Computational Complexity) Algorithm 3 solves Problem 3 over $\mathbb{F}_{q^m}[x; \sigma]^{s+1}$ in $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

4.3 Interpolation-based decoding of interleaved skew Reed–Solomon codes

In this section we consider decoding of Skew Reed–Solomon (SRS) codes with respect to the *skew metric*, which was introduced in [39]. Decoding schemes for SRS codes that allow for correcting error of skew weight up to $\lfloor \frac{n-k}{2} \rfloor$ were presented in [11, 13, 34, 39]. Interpolation-based decoding of *interleaved* SRS ISRS codes that allows for decoding errors of skew weight up to $\frac{s}{s+1}(n-k+1)$, where *s* is the *interleaving order*, was considered in [6]. The interpolation scheme can be either used as a list decoder or as a probabilistic-unique decoder.

In this work we consider the definition of the skew weight from [13, Proposition 1]. Let the vector $\mathbf{b} \in \mathbb{F}_{q^m}^n$ contain *P*-independent elements and let \mathcal{B} be the *P*-closed set generated by the elements in **b**. Then, the *skew weight* of a vector $\mathbf{x} \in \mathbb{F}_{q^m}^n$ with respect to \mathcal{B} is defined as

$$\operatorname{wt}_{skew}(\mathbf{x}) := \operatorname{deg}\left(\operatorname{lclm}\left(x - b_i^{x_i}\right)_{\substack{0 \le i \le n-1 \\ x_i \ne 0}}\right)$$
(73)

where $b_i{}^{x_i} := \sigma(x_i) b_i x_i^{-1} + \delta(x_i) x_i^{-1}$ (see (3).) For simplicity we omit the dependence on \mathcal{B} in the definition of $\operatorname{wt}_{skew}(\cdot)$ as it will be clear from the context. Similar to the rank and the sum-rank weight we have that $\operatorname{wt}_{skew}(\mathbf{x}) \le \operatorname{wt}_H(\mathbf{x})$ for all $\mathbf{x} \in \mathbb{F}_{q^m}^n$ (see [39]). The *skew distance* between two vectors $\mathbf{x}, \mathbf{y} \in \mathbb{F}_{q^m}^n$ is defined as

$$d_{skew}(\mathbf{x}, \mathbf{y}) := \mathrm{wt}_{skew}(\mathbf{x} - \mathbf{y}). \tag{74}$$

As a channel model we consider the skew metric channel

$$\mathbf{r} = \mathbf{c} + \mathbf{e} \tag{75}$$

where the error vector **e** has skew weight $wt_{skew}(\mathbf{e}) = t$.

We define (vertically) interleaved skew Reed-Solomon (ISRS) codes as follows.

Definition 10 (Interleaved Skew Reed–Solomon Code) Let $\mathbf{b} = (b_0, b_1, \dots, b_{n-1}) \in \mathbb{F}_{q^m}^n$ contain *P*-independent elements from \mathbb{F}_{q^m} . For a fixed integer $k \leq n$, a vertically *s*-interleaved skew Reed–Solomon (ISRS) code of length *n* and dimension *k* is defined as

$$\operatorname{ISRS}[\mathbf{b}, s; n, k] = \left\{ \begin{pmatrix} f^{(1)}[\mathbf{b}] \\ f^{(2)}[\mathbf{b}] \\ \vdots \\ f^{(s)}[\mathbf{b}] \end{pmatrix} : f^{(j)} \in \mathbb{F}_{q^m}[x; \sigma]_{< k}, \forall j \in [1, s] \right\}.$$
(76)

ISRS codes are maximum skew distance (MSD) codes, i.e. we have that

$$d_{skew}(\text{ISRS}[\mathbf{b}, s; n, k]) = n - k + 1.$$
(77)

Under a fixed basis of $\mathbb{F}_{q^{ms}}$ over \mathbb{F}_{q^m} there is a bijection between a matrix $\mathbf{C} \in \mathbb{F}_{q^m}^{s \times n}$ and a vector $\mathbf{c} \in \mathbb{F}_{q^{ms}}^n$. Hence, we can represent each codeword $\mathbf{C} \in \text{ISRS}[\mathbf{b}, s; n, k]$ as a vector $\mathbf{c} = f(\mathbf{b}) \in \mathbb{F}_{q^{ms}}^n$ where $f \in \mathbb{F}_{q^{ms}}[\mathbf{x}; \sigma]$ is the the polynomial obtained by considering the coefficients of $f^{(1)}, \ldots, f^{(s)}$ over $\mathbb{F}_{q^{ms}}$. This relation between *s*-interleaved evaluation codes over \mathbb{F}_{q^m} and punctured evaluation codes over the bigger field $\mathbb{F}_{q^{ms}}$ is well-known from Reed–Solomon and Gabidulin codes (see e.g. [8, 58]).

Suppose we transmit a vector $\mathbf{c} = f(\mathbf{b}) \in \text{ISRS}[\mathbf{b}, s; n, k]$ and receive a vector

$$\mathbf{r} = \mathbf{c} + \mathbf{e} \in \mathbb{F}_{q^{ms}}^n \tag{78}$$

where wt_{skew}(\mathbf{e}) = t. Let $\mathbf{R} \in \mathbb{F}_{q^m}^{s \times n}$ denote the expanded matrix of $\mathbf{r} \in \mathbb{F}_{q^{ms}}^n$ over \mathbb{F}_{q^m} .

Definition 11 (*Remainder Vector Evaluation Map*) Given an interpolation point set $\mathcal{P} = \{(p_{i,0}, p_{i,1}, \dots, p_{i,s}) : i = 0, \dots, n-1\} \subseteq \mathbb{F}_{q^m}^{s+1}$ and a vector $\mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ we define the remainder vector evaluation maps as

$$\mathscr{E}_{i}^{rem}(\mathbf{Q}) := \mathcal{Q}_{0}[p_{i,0}] + \sum_{j=1}^{s} \mathcal{Q}_{j}[p_{i,0}^{p_{i,j}}] p_{i,j}, \quad \forall i = 0, \dots, n-1.$$
(79)

Remark 2 Note that $p_{i,0}p_{i,j}$ is not defined whenever $p_{i,j} = 0$. Similar to [43] we define $Q_j[p_{i,0}p_{i,j}]p_{i,j} = 0$ for all $p_{i,j} = 0$. This definition is very natural in view of the correspondence to the generalized operator evaluation (see Lemma 3) and the product rule (see [29, Theorem 2.7]).

For interleaved skew Reed-Solomon codes, the interpolation point set is

$$\mathcal{P} = \{ (\beta_i, r_{0,i}, \dots, r_{s-1,i}) : i = 0, \dots, n-1 \} \subset \mathbb{F}_{a^m}^{s+1}.$$
(80)

Problem 4 (Vector Interpolation Problem) Given the integer $s \in \mathbb{Z}_+$, a degree constraint $D \in \mathbb{Z}_+$, a set of \mathbb{F}_{q^m} -linear vector evaluation maps $\mathcal{E}^{rem} = \{\mathscr{E}_0^{rem}, \ldots, \mathscr{E}_{n-1}^{rem}\}$ as defined in (79) and a vector $\mathbf{w} = (0, k - 1, \ldots, k - 1) \in \mathbb{Z}_+^{s+1}$ compute a vector $\mathbf{b} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ that satisfies:

$$-\mathscr{E}_{i}^{rem}(\mathbf{b})=0,\,\forall i=0,\ldots,n-1,$$

$$-\deg_{\mathbf{w}}(\mathbf{b}) < D.$$

A nonzero solution of Problem 4 exists if the degree constraint satisfies (see [6])

$$D = \left\lceil \frac{n + s(k-1) + 1}{s+1} \right\rceil.$$
(81)

Proposition 3 Problem 4 can be solved by Algorithm 1 in $O(s^2n^2)$ operations in \mathbb{F}_{q^m} .

Proof Problem 4 corresponds to finding a vector in the interpolation module

$$\overline{\mathcal{K}}_{n-1} = \{ \mathbf{b} \in \mathbb{F}_{q^m}[x;\sigma]^{s+1} : \mathscr{E}_i^{rem}(\mathbf{b}) = 0, \forall i = 0, \dots, n-1 \}$$
(82)

of minimal **w**-weighted degree. Hence, Problem 4 can be solved by computing a **w**-ordered weak-Popov basis for the left $\mathbb{F}_{q^m}[x; \sigma]$ -module $\overline{\mathcal{K}}_{n-1}$ and returning a row with **w**-weighted degree less than *D*. The problem of finding a **w**-ordered weak-Popov basis for $\overline{\mathcal{K}}_{n-1}$ is an instance of the general vector interpolation problem (Problem 1) which can be solved by Algorithm 1 (see Lemma 4).

By Proposition 1, Algorithm 1 requires the computation of O(sn) evaluation maps \mathscr{E}_i^{rem} of a vector in $\mathbb{F}_{q^m}[x;\sigma]_{\leq n}^{s+1}$, which requires O(sn) operations in \mathbb{F}_{q^m} each. Overall, the computation of the evaluation maps requires $O(s^2n^2)$ operations in \mathbb{F}_{q^m} . The computation of the *n* multiplications of a monic degree-1 skew polynomial with a vector from $\mathbb{F}_{q^m}[x;\sigma]$ (Line 11) requires $O(sn^2)$ operations in \mathbb{F}_{q^m} in total. The O(sn) multiplications of an element from \mathbb{F}_{q^m} and a vector from $\mathbb{F}_{q^m}[x;\sigma]_{\leq n}^{s+1}$ require at most $O(s^2n^2)$ operations in \mathbb{F}_{q^m} . Therefore we conclude that Algorithm 1 can solve Problem 2 requiring at most $O(s^2n^2)$ operations in \mathbb{F}_{q^m} .

If the skew weight of the error $t = wt_{skew}(\mathbf{e})$ satisfies

$$t < \frac{s}{s+1}(n-k+1)$$
(83)

it can be shown that (see [6])

$$P(x) = Q_0(x) + \sum_{j=1}^{s} Q_j(x) f^{(j)}(x) = 0.$$
(84)

The root-finding step consists of finding all polynomials $f^{(1)}, \ldots, f^{(s)} \in \mathbb{F}_{q^m}[x; \sigma]_{<k}$ that satisfy (84). The root-finding problem can be solved efficiently by the minimal approximant basis methods in [10, 11] requiring at most $\tilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

4.3.1 Solving the interpolation step via the fast KNH interpolation

Define the sets

$$\mathcal{P}_{[i,j]}^{(r)} := \left\{ p_{l,0}^{p_{l,r}} : p_{l,r} \neq 0, \forall l = i, \dots, j \right\}$$
(85)

for all $0 \le i, j \le n - 1$ and r = 1, ..., s. Then the vectors of minimal polynomials with respect to the generalized operator evaluation are defined as

$$\mathbf{M}_{[i,j]}^{\text{rem}}(x) := \left(M_{\{p_{i,0},\dots,p_{j,0}\}}^{\text{rem}}(x), M_{\mathcal{P}_{[i,j]}^{(1)}}^{\text{rem}}(x),\dots, M_{\mathcal{P}_{[i,j]}^{(s)}}^{\text{rem}}(x) \right).$$
(86)

🖄 Springer

Lemma 10 Let $\mathcal{E}^{rem} = (\mathscr{E}_0^{rem}, \ldots, \mathscr{E}_{n-1}^{rem})$ be an ordered set of skew vector evaluation maps as defined in (79) and let $\mathcal{E}_{[i,j]}^{rem} = (\mathscr{E}_i^{rem}, \ldots, \mathscr{E}_j^{rem}) \subseteq \mathcal{E}^{rem}$. Then for any $\mathbf{Q} \in \mathbb{F}_{q^m}[x; \sigma]^{s+1}$ we have that

$$\mathscr{E}_{l}^{rem}(\mathbf{Q}) = \mathscr{E}_{l}^{rem}(\mathbf{Q} \bmod_{\mathbf{r}} \mathbf{M}_{[i,j]}^{rem}(x)), \quad \forall l = i, \dots, j.$$
(87)

Proof For the case where the interpolation point \mathbf{p}_i (corresponding to \mathscr{E}_i^{rem}) contains only nonzero elements, the lemma follows directly by applying the result from Lemma 2 to the elementary evaluations in the skew vector remainder evaluation maps defined in (79). Since by definition of the sets in (85) all conjugates $p_{i,0}^{p_{i,r}}$ where $p_{i,r} = 0$ are excluded, and by definition the evaluation of each $Q_r[p_{i,0}^{p_{i,r}}]p_{i,r} = 0$ for all $p_{i,r} = 0$ (see Remark 2), we have that the statement also holds in this case.

An important consequence of Lemma 10 is, that the vector remainder evaluation maps from Definition 11 and the minimal polynomial vectors in (86) fulfill Assumption 1. Hence, we can solve Problem 4 by calling Algorithm 3 with SkewInterpolateTree(\mathcal{E}^{rem} , \mathbf{I}_{s+1} , (0, k - 1, ..., k - 1)).

4.3.2 Complexity analysis

Lemma 11 (Complexity of Computing Minimal Polynomial Vectors) *Algorithm 4 constructs the (ordered) set containing the minimal polynomial vectors*

$$\mathcal{M}^{rem} := \left(\mathbf{M}_{[0,n-1]}^{rem}(x), \mathbf{M}_{[0,\lfloor n/2 \rfloor - 1]}^{rem}(x), \mathbf{M}_{[\lfloor n/2 \rfloor, n-1]}^{rem}(x) \dots, \mathbf{M}_{[n-1,n-1]}^{rem}(x) \right) \\ \subset \mathbb{F}_{q^m}[x;\sigma]^{s+1}$$
(88)

as defined in (86) in \widetilde{O} (s $\mathfrak{M}(n)$) operations in \mathbb{F}_{q^m} .

Proof Algorithm 4 is a generalization of the procedure in [17, Theorem 3.2.7] to construct a single minimal polynomial, which requires $\tilde{O}(\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} . Hence, the overall complexity of Algorithm 4 is in the order of $\tilde{O}(s\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

Theorem 3 (Computational Complexity) Algorithm 3 solves Problem 4 over $\mathbb{F}_{q^m}[x; \sigma]^{s+1}$ in $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} .

Proof Follows the ideas of the proof of Theorem 2.

Remark 3 (Practical Consideration) Note that in the $\mathbb{F}_{q^m}[x; \sigma]$ -case the *i*-th remainder vector evaluation map $\mathscr{E}_i^{rem}(x\mathbf{Q})$ of $x\mathbf{Q}$ can be computed efficiently from $\mathscr{E}_i^{rem}(\mathbf{Q})$ requiring only one application of the automorphism σ and one multiplication since by the product rule [29] we have that

$$\mathscr{E}_{i}^{rem}(x\mathbf{Q}) = \sigma\left(\mathscr{E}_{i}^{rem}(\mathbf{Q})\right) p_{i,0}, \quad \forall i = 0, \dots, n-1.$$
(89)

4.4 Applications to other related coding problems

Algorithm 3 with generalized operator vector evaluation maps of the form as defined in Definition 8 can be used to solve the interpolation step in other related interpolation-based decoding problems efficiently. The corresponding interpolation problems are instances of Problem 1 where the minimal polynomial vectors in (55) fulfill Assumption 1.

In particular, the interpolation step in decoding of lifted *s*-interleaved Gabidulin codes in the subspace metric [8, 9] can be performed requiring at most $\widetilde{O}(s^{\omega}\mathfrak{M}(n_r))$ operations in \mathbb{F}_{q^m} , where n_r denotes the dimension of the received subspace. Further, the interpolation step in decoding *h*-folded Gabidulin codes [8, 37] and *lifted h*-folded Gabidulin codes [7] can be performed in at most $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ and $\widetilde{O}(s^{\omega}\mathfrak{M}(n_r))$ operations in \mathbb{F}_{q^m} , respectively, where s < h denotes an interpolation parameter.

Algorithm 3 can also be used to solve the interpolation step for decoding lifted *s*-interleaved LRS codes in the sum-subspace metric [5] requiring at most $\tilde{O}(s^{\omega}\mathfrak{M}(n_r))$ operations in \mathbb{F}_{q^m} , where n_r denotes the sum of the dimensions of the received subspaces over all shots.

Algorithm 3 can also solve the interpolation step for interpolation-based decoding of *h*-folded lLRS codes [23] requiring at most $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} , where $s \leq h$ is a decoding parameter and *n* is the length of the unfolded code.

Since, unlike the minimal approximant basis approach in [6, 11], Algorithm 3 has no requirements on the interpolation points (related to the evaluation maps), it can be applied in a straight-forward manner to decoding problems of lifted Gabidulin and LRS code variants in the (sum-)subspace metric.

5 Conclusion

We proposed a fast divide-and conquer variant of the Kötter-Nielsen-Høholdt (KNH) interpolation over free modules over skew polynomial rings. We showed how the proposed KNH interpolation can be used to solve the interpolation step of interpolation-based decoding of interleaved Gabidulin, linearized Reed-Solomon and skew Reed-Solomon codes and variants thereof efficiently requiring at most $\widetilde{O}(s^{\omega}\mathfrak{M}(n))$ operations in \mathbb{F}_{q^m} , where n is the length of the code, s the interleaving order, $\mathfrak{M}(n)$ the complexity for multiplying two skew polynomials of degree at most n, ω the matrix multiplication exponent and O (·) the soft-O notation which neglects log factors. The computational complexity of the proposed fast KNH variant coincides with the complexity of the currently fastest interpolation algorithms for skew polynomial rings, where the proposed variant relies on the well-known (bottom-up) KNH interpolation algorithm instead of quite involved (top-down) minimal approximant bases techniques. The proposed results also hold for codes defined over general skew polynomials rings (with derivations), except for the complexity analysis. Due to the bottom-up nature of the proposed KNH interpolation there are no requirements on the interpolation points and thus no pre-processing of the interpolation points, which may be required for the top-down minimal approximant bases approaches, is necessary.

Acknowledgements The authors would like to thank Felicitas Hörmann for fruitful discussions on earlier versions of the paper and the anonymous reviewers for their valuable comments that helped to improve the presentation of the paper.

Funding Open Access funding enabled and organized by Projekt DEAL.

Data Availability No datasets have been used.

Declarations

Conflict of interest Nil.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Adams W.W., Loustaunau P.: An Introduction to Gröbner Bases, vol. 3. American Mathematical Society, Providence (1994).
- Alekhnovich M.: Linear Diophantine equations over polynomials and soft decoding of Reed-Solomon codes. In: The 43rd Annual IEEE Symposium on Foundations of Computer Science, 2002. Proceedings, pp. 439–448. IEEE (2002).
- Alman J., Williams V.V.: A refined laser method and faster matrix multiplication. In: Proceedings of the 2021 ACM-SIAM Symposium on Discrete Algorithms (SODA), pp. 522–539. SIAM (2021).
- Augot D., Couvreur A., Lavauzelle J., Neri A.: Rank-metric codes over arbitrary Galois extensions and rank analogues of Reed-Muller codes. SIAM J. Appl. Algebra Geom. 5(2), 165–199 (2021).
- Bartz H., Puchinger S.: Decoding of interleaved linearized Reed-Solomon codes with applications to network coding. In: IEEE International Symposium on Information Theory (ISIT). IEEE (2021).
- Bartz H., Puchinger S.: Fast decoding of interleaved linearized Reed–Solomon codes and variants. IEEE Trans. Inf. Theory (2022). arXiv:2201.01339.
- Bartz H., Sidorenko V.: List and probabilistic unique decoding of folded subspace codes. In: 2015 IEEE International Symposium on Information Theory (ISIT), pp. 11–15. IEEE (2015).
- Bartz H., Sidorenko V.: Algebraic decoding of folded Gabidulin codes. Des. Codes Cryptogr. 82(1), 449–467 (2017).
- Bartz H., Wachter-Zeh A.: Efficient decoding of interleaved subspace and Gabidulin codes beyond their unique decoding radius using Gröbner bases. Adv. Math. Commun. 12(4), 773 (2018).
- Bartz H., Jerkovits T., Puchinger S., Rosenkilde J.: Fast root finding for interpolation-based decoding of interleaved Gabidulin codes. In: 2019 IEEE Information Theory Workshop (ITW), pp. 1–5. IEEE (2019).
- 11. Bartz H., Jerkovits T., Puchinger S., Rosenkilde J.: Fast decoding of codes in the rank, subspace, and sum-rank metric. IEEE Trans. Inf. Theory **67**(8), 5026–5050 (2021).
- Bartz H., Jerkovits T., Rosenkilde J.: Fast Kötter–Nielsen–H
 øholdt interpolation over skew polynomial rings. In: 25th International Symposium on Mathematical Theory of Networks and Systems (MTNS) (2022).
- Boucher D.: An algorithm for decoding skew Reed-Solomon codes with respect to the skew metric. Des. Codes Cryptogr. 88(9), 1991–2005 (2018).
- Boucher D., Ulmer F.: Linear codes using skew polynomials with automorphisms and derivations. Des. Codes Cryptogr. 70(3), 405–431 (2014).
- 15. Caruso X.: Residues of skew rational functions and linearized Goppa codes (2019). arXiv:1908.08430.
- Caruso X., Le Borgne J.: Fast multiplication for skew polynomials. In: International Symposium on Symbolic and Algebraic Computation (ISSAC) (2017).
- Caruso X., Le Borgne J.: A new faster algorithm for factoring skew polynomials over finite fields. J. Symb. Comput. 79, 411–443 (2017).
- 18. Cox D., Little J., O'Shea D.: Ideals, Varieties, and Algorithms, vol. 3. Springer, Berlin (1992).
- 19. Fitzpatrick P.: On the key equation. IEEE Trans. Inf. Theory 41(5), 1290–1302 (1995).
- 20. Gabidulin E.M.: Theory of codes with maximum rank distance. Probl. Inf. Transm. 21(1), 3–16 (1985).
- 21. Geiselmann W., Ulmer F.: Skew Reed Muller Codes, pp. 107–116. HAL Open Science, Lyon (2019).
- Giorgi P., Jeannerod C.P., Villard G.: On the complexity of polynomial matrix computations. In: Proceedings of the 2003 International Symposium on Symbolic and Algebraic Computation, pp. 135–142 (2003).
- Hörmann F., Bartz H.: Efficient decoding of folded linearized Reed-Solomon codes in the sum-rank metric (2021). arXiv:2109.14943.
- Jeannerod C.P., Neiger V., Schost É., Villard G.: Computing minimal interpolation bases. J. Symb. Comput. 83, 272–314 (2017).

- Kojima C., Rapisarda P., Takaba K.: Canonical forms for polynomial and quadratic differential operators. Syst. Control Lett. 56(11–12), 678–684 (2007).
- Kötter R.: On algebraic decoding of algebraic-geometric and cyclic codes. Dissertation, Linköping University of Technology (1996).
- 27. Lam T.Y.: A General Theory of Vandermonde Matrices. Center for Pure and Applied Mathematics, University of California, Berkeley (1985).
- Lam T., Leroy A.: Algebraic conjugacy classes and skew polynomial rings. In: Perspectives in Ring Theory, pp. 153–203. Springer, Berlin (1988).
- Lam T.Y., Leroy A.: Vandermonde and Wronskian matrices over division rings. J. Algebra 119(2), 308– 336 (1988).
- Lam T., Leroy A.: Hilbert 90 theorems over division rings. Trans. Am. Math. Soc. 345(2), 595–622 (1994).
- Leroy A., et al.: Pseudolinear transformations and evaluation in ore extensions. Bull. Belg. Math. Soc. Simon Stevin 2(3), 321–347 (1995).
- Liu S.: Generalized skew Reed-Solomon codes and other applications of skew polynomial evaluation. Ph.D. thesis, University of Toronto (Canada) (2016).
- Liu S., Manganiello F., Kschischang F.R.: Kötter interpolation in skew polynomial rings. Des. Codes Cryptogr. 72(3), 593–608 (2014).
- Liu S., Manganiello F., Kschischang F.R.: Construction and decoding of generalized skew-evaluation codes. In: 2015 IEEE 14th Canadian Workshop on Information Theory (CWIT), pp. 9–13. IEEE (2015).
- Loidreau P., Overbeck R.: Decoding rank errors beyond the error correcting capability. In: International Workshop Algebraic and Combinatorial Coding Theory (ACCT) (2006).
- Lu H.F., Kumar P.V.: A unified construction of space-time codes with optimal rate-diversity tradeoff. IEEE Trans. Inf. Theory 51(5), 1709–1730 (2005).
- Mahdavifar H., Vardy A.: List-decoding of subspace codes and rank-metric codes up to singleton bound. In: 2012 IEEE International Symposium on Information Theory Proceedings, pp. 1488–1492. IEEE (2012).
- Martinez-Penas U.: On the similarities between generalized rank and hamming weights and their applications to network coding. IEEE Trans. Inf. Theory 62(7), 4081–4095 (2016).
- Martínez-Peñas U.: Skew and linearized Reed-Solomon codes and maximum sum rank distance codes over any division ring. J. Algebra 504, 587–612 (2018).
- Martínez-Peñas U.: Private information retrieval from locally repairable databases with colluding servers. In: 2019 IEEE International Symposium on Information Theory (ISIT), pp. 1057–1061. IEEE (2019).
- Martínez-Peñas U.: Theory and applications of linearized multivariate skew polynomials. Linear Algebra Appl. 637, 1–23 (2022).
- Martínez-Peñas U., Kschischang F.R.: Evaluation and interpolation over multivariate skew polynomial rings. J. Algebra 525, 111–139 (2019).
- Martínez-Peñas U., Kschischang F.R.: Reliable and secure multishot network coding using linearized Reed-Solomon codes. IEEE Trans. Inf. Theory 65(8), 4785–4803 (2019).
- Martínez-Peñas U., Kschischang F.R.: Universal and dynamic locally repairable codes with maximal recoverability via sum-rank codes. IEEE Trans. Inf. Theory 65(12), 7790–7805 (2019).
- Middeke J.: A computational view on normal forms of matrices of ore polynomials. Ph.D. thesis, Johannes Kepler University Linz (2011).
- Mulders T., Storjohann A.: On lattice reduction for polynomial matrices. J. Symb. Comput. 35(4), 377–401 (2003).
- Neiger V.: Bases of relations in one or several variables: fast algorithms and applications. Ph.D. thesis, École Normale Supérieure de Lyon - University of Waterloo (2016).
- Nielsen J.S.R.: List Decoding of Algebraic Codes. Technical University of Denmark, Kongens Lyngby (2013).
- Nielsen J.S.: Fast Kötter-Nielsen-H
 øholdt interpolation in the Guruswami-Sudan algorithm. In: 14th International Workshop on Algebraic and Combinatorial Coding Theory (ACCT) (2014).
- Nielsen R.R., Høholdt T.: Decoding Reed-Solomon codes beyond half the minimum distance. In: Coding Theory, Cryptography and Related Areas, pp. 221–236. Springer (2000).
- Nóbrega R.W., Uchôa-Filho B.F.: Multishot codes for network coding using rank-metric codes. In: 2010 Third IEEE International Workshop on Wireless Network Coding, pp. 1–6. IEEE (2010).
- 52. Ore Ø.: On a special class of polynomials. Trans. Am. Math. Soc. 35, 559–584 (1933).
- 53. Ore O.: Theory of non-commutative polynomials. Ann. Math. 34, 480–508 (1933).
- Overbeck R.: Decoding interleaved Gabidulin codes and ciphertext-security for GPT variants. IACR Cryptol. ePrint Arch. 2006, 222 (2006).

- Puchinger S., Wachter-Zeh A.: Fast operations on linearized polynomials and their applications in coding theory. J. Symb. Comput. 89, 194–215 (2018).
- Puchinger S., Renner J., Rosenkilde J.: Generic decoding in the sum-rank metric. IEEE Trans. Inf. Theory 68(8), 5075–5097 (2022).
- Sidorenko V., Bossert M.: Decoding interleaved Gabidulin codes and multisequence linearized shiftregister synthesis. In: 2010 IEEE International Symposium on Information Theory, pp. 1148–1152. IEEE (2010).
- Sidorenko V., Schmidt G., Bossert M.: Decoding punctured Reed–Solomon codes up to the singleton bound. In: 7th International ITG Conference on Source and Channel Coding, pp. 1–6. VDE (2008).
- Sidorenko V., Jiang L., Bossert M.: Skew-feedback shift-register synthesis and decoding interleaved Gabidulin codes. IEEE Trans. Inf. Theory 57(2), 621–632 (2011).
- 60. Sturmfels B.: What is a Gröbner basis. Not. Am. Math. Soc. 52(10), 1199-1200 (2005).
- Sudan M.: Decoding of Reed Solomon codes beyond the error-correction bound. J. Complex. 13(1), 180–193 (1997).
- Wachter-Zeh A., Zeh A.: List and unique error-erasure decoding of interleaved Gabidulin codes with interpolation techniques. Des. Codes Cryptogr. 73(2), 547–570 (2014).
- Wang B., McEliece R.J., Watanabe K.: Kötter interpolation over free modules. In: Proceedings of, pp. 2197–2206 (2005).
- 64. Welch L.R., Berlekamp E.R.: Error correction for algebraic block codes. US Patent 4,633,470 (1986).
- Xie H., Yan Z., Suter B.W.: General linearized polynomial interpolation and its applications. In: 2011 International Symposium on Networking Coding, pp. 1–4. IEEE (2011).

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.